UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT

April 18, 2014

Table of Contents

S	TATE	MENT OF PURPOSE	5
1	. KL	AMATH ADJUDICATION	5
	1.1. and E	Provisional Settlement of Certain Klamath Adjudication Tribal Water Rights Clain acceptions.	
	1.2.	Agreement to Not File Exceptions Regarding Certain Irrigation Claims.	6
	1.3.	Process for Resolution of Additional Claims and Exceptions.	6
	1.4.	Deadline for Filing of Exceptions.	6
	1.5.	Deferral of Proceedings on Exceptions to Certain Claims	7
	1.6.	Conditional Resolution of Off-Reservation Claims	7
	1.7.	Adjudication Parties as Agreement Parties.	
2.	EC	ONOMIC DEVELOPMENT	9
	2.1.	Economic Self-Sufficiency.	9
	2.2.	Mazama Forest.	9
	2.3.	Water Use and Riparian Programs	9
	2.4.	Economic Self-Sufficiency.	9
	2.5. Right	Klamath Tribes' Relinquishment of Claims to the United States and Retention of	9
	2.6.	Jobs Program.	. 11
3.	V	VATER USE PROGRAM (WUP)	. 12
	3.1.	Overall Administration of the Water Use Program	. 12
	3.2.	Water Use Program Outcomes.	. 12
	3.3.	Increased Inflow into Upper Klamath Lake.	. 12
	3.4.	Other Water Management Changes.	. 13
	3.5.	WUP Guidelines.	.13
	3.6.	Water Use Agreements.	.15
	3.7.	Tracking WUP Region Volumes.	16
	3.8.	Use of the WUP Ledger.	18
	3.9.	Specified Instream Flow (SIF) Call Thresholds.	18
	3.10.	Regulatory Calls Resulting from Nonperformance Under the Riparian Program	21
	3.11.	Groundwater Regulation.	23
	3.12.	Means of Achieving WUP Volumes and Call Thresholds.	29
	3.13.	Monitoring of Alternate Methods.	30
	3.14.	Water Use Agreements	30

	3.15.	Willing Sellers.	
	3.16.	Water Banking.	30
	3.17.	Relationship Between Water Use Agreements and Riparian Management	
		ements.	
	3.18.	Limitations on the Retirement of Water Rights	
	3.19.	Practicably Irrigable Acres and Other Water Rights that Are Not Fully Developed	
	3.20.	Dewatering	
	3.21.	Livestock Watering	
	3.22.	Preliminary Direction to the JME and LE.	. 32
	3.23.	Klamath Reclamation Project.	. 32
	3.24.	Alteration of Law.	
	3.25.	Alteration of Procedures.	
4.	. RIP	ARIAN PROGRAM	. 34
	4.1.	Overall Administration of the Riparian Program.	. 34
	4.2.	Riparian Program Outcomes.	. 34
	4.3.	Categories of Areas in the Riparian Program.	. 35
	4.4.	Monitoring and Adaptive Management in the Three Categories of Areas	. 35
	4.5.	Riparian Management Agreements and Riparian Action Plan.	. 36
	4.6.	Approval of Agreements.	41
	4.7.	Amendment of Riparian Guidelines	41
	4.8.	Sufficient Participation.	41
	4.9.	Deviation from Sufficient Participation Criterion.	41
	4.10.	Permanence of Riparian Management Agreements and Transfer of Ownership	42
	4.11.	Relation between the Riparian Program and the Water Use Program	42
	4.12.	RMA Ledger.	42
5.	TRA	ANSITION PERIOD	43
	5.1.	Introduction	43
	5.2.	Initial Administration	43
	5.3.	Transitional Water Use Program	44
	5.4.	Transitional Riparian Program	45
	5.5.	Regulation of Water Use During the Transition Period.	46
5.		DITIONAL AGREEMENTS	
	6.1.	Access for Exercise of Treaty Rights	48
			48

	6.3.	Klamath Allottees.	48
	6.4.	Walton Rights.	48
7.	JOI	NT MANAGEMENT ENTITY	49
	7.1.	Formation.	
	7.2.	Functions.	50
	7.3.	Decision-making.	51
	7.4.	Dispute Resolution.	51
	7.5.	Articles of Incorporation.	52
	7.6.	Initial Funding.	52
	7.7.	Permanent Funding.	52
	7.8.	JME Technical Team.	52
	7.9.	Coordination Prior to Formation of the JME.	52
8.	LAì	NDOWNER ENTITY	54
	8.1.	Formation.	54
	8.2.	Powers and Duties - Water Use Program.	
	8.3.	Powers and Duties – Riparian Program.	55
	8.4.	Other Powers of the LE.	55
	8.5.	Funding.	55
9.	REC	GULATORY ASSURANCES	57
1(). C	ONTINGENCIES	61
	10.1.	Secretarial Affirmative Notice.	61
	10.2.	Secretarial Negative Notice.	63
	10.3	Judicial Review of Secretarial Finding.	63
	10.4 Settled	Common Litigation Strategy Regarding Non-Party Exceptions to the Provisionally Tribal Water Right Claims	
	10.5.	Court Proceedings Prior to Secretarial Affirmative Notice.	65
11	l. M	EET AND CONFER	66
	11.1.	Procedure.	66
	11.2.	Non-Binding Mediation.	66
12	2. G	ENERAL TERMS	67
	12.1.	Amendments	67
	12.2.	Obligations Under This Agreement.	67
	12.3.	Support for Necessary Federal Authorizing Legislation.	70
	12.4.	Support for Necessary Oregon Authorizing Legislation.	71

12.5.	Force Majeure.	72
12.6.	Untimely or Inadequate Funding.	72
12.7.	Severability.	73
12.8.	Changed Circumstances.	73
12.9.	Federal Agencies as Parties.	73
12.10). Enforcement	74
12.11	. Successors and Assigns.	75
12.12	. Joint Venture.	75
12.13	. Governing Law.	75
12.14	. Elected Officials Not to Benefit	76
12.15	. Entire Understanding.	76
10.16		7.0
12.16	. Treatment of Communications Related to Agreement.	/6
	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE	
13L 7	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE	NTS.
13. L 7' 13.1.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE 7	NTS. 77
13. L 7' 13.1. 13.2.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE 7 Compliance with Legal Responsibilities.	NTS. 77
13. L 7' 13.1. 13.2. 13.3.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE 7 Compliance with Legal Responsibilities. Reservation of Rights.	NTS 77 77 80
13. L 7' 13.1. 13.2. 13.3. 14. EXE	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE 7 Compliance with Legal Responsibilities. Reservation of Rights. Calls for Regulation.	NTS 77 77 80 81
13. L 7' 13.1. 13.2. 13.3. 14. EXE 14.1.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE Compliance with Legal Responsibilities. Reservation of Rights. Calls for Regulation. CUTION OF THE AGREEMENT.	NTS 77 80 81 81
13. L 7' 13.1. 13.2. 13.3. 14. EXE 14.1. 14.2.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE Compliance with Legal Responsibilities. Reservation of Rights. Calls for Regulation. CUTION OF THE AGREEMENT. Authority.	NTS 77 80 81 81
13. L 7' 13.1. 13.2. 13.3. 14. EXE 14.1. 14.2. 14.3.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE Compliance with Legal Responsibilities. Reservation of Rights. Calls for Regulation. CUTION OF THE AGREEMENT. Authority. Counterparts.	NTS 77 80 81 81 81 81
13. L 7' 13.1. 13.2. 13.3. 14. EXE 14.1. 14.2. 14.3.	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE Compliance with Legal Responsibilities. Reservation of Rights. Calls for Regulation. CUTION OF THE AGREEMENT. Authority. Counterparts. Effective Date.	NTS 77 80 81 81 81 81 82
13. L 7' 13.1. 13.2. 13.3. 14. EXE 14.1. 14.2. 14.3. 15. DEF	EGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDE Compliance with Legal Responsibilities. Reservation of Rights. Calls for Regulation. CUTION OF THE AGREEMENT. Authority. Counterparts. Effective Date. INITIONS.	NTS 77 80 81 81 81 81 82 91

STATEMENT OF PURPOSE

The Purposes of this Upper Klamath Basin Comprehensive Agreement ("Agreement") are to achieve four co-equal goals:

- (a) To support the economic development interests of the Klamath Tribes;
- (b) To provide a stable, sustainable basis for the continuation of agriculture in the Upper Klamath Basin;
- (c) To manage and restore riparian corridors along streams that flow into Upper Klamath Lake in order to achieve Proper Functioning Conditions permanently; and
- (d) To resolve controversies regarding certain water right claims, contests, and exceptions in the ongoing Klamath Adjudication in the Klamath County Circuit Court.

NOW, THEREFORE, the Parties agree as follows:

1. KLAMATH ADJUDICATION.

Subject to subsection 1.5, the Adjudication Parties agree to use Best Efforts to avoid as much of the litigation in the Klamath Adjudication on the claims listed in subsections 1.1 through 1.3 as possible.

1.1. Provisional Settlement of Certain Klamath Adjudication Tribal Water Rights Claims and Exceptions. The Adjudication Parties agree that they will withdraw any exceptions, responses to exceptions, and/or replies to responses to exceptions that they have filed concerning the Provisionally Settled Tribal Water Right Claims within thirty (30) days after the date of the publication of the Affirmative Notice provided for in subsection 10.1 that makes this Agreement permanent.

The "Provisionally Settled Tribal Water Right Claims" means the following water right claims of the Klamath Tribes and the United States Bureau of Indian Affairs (BIA) in its capacity as trustee for the Klamath Tribes: claims 616 and 622 (Upper Klamath Lake); claims 625-630, 634, and 640 and the relevant portion of claim 612 (Williamson River Basin instream flow claims below Klamath Marsh); claims 641-657 and the relevant portion of claim 612 (Sprague River Basin instream flow claims); claims 658-662 and the relevant portion of claim 612 (Sycan River Basin instream flow claims below Sycan Marsh); and claims 668-670 and the relevant portion of claim 612 (Wood River

Basin instream flow claims), and any claims that are subsequently settled under subsection 1.3.1.

- 1.2. Agreement to Not File Exceptions Regarding Certain Irrigation Claims. The United States and the Klamath Tribes agree to not file exceptions in the Klamath County Circuit Court concerning certain Klamath Adjudication claims that have been previously settled between individual irrigators and the United States. Those claim numbers are: 46, 51, 53, 59, 61, 62, 63, 64, 66, 67, 76, 84, 98, 99, 108, 109, 113, 119, 120, 121, 122, 128, 129, 130, 234, 270, 279, and 697.
- 1.3. Process for Resolution of Additional Claims and Exceptions. The Adjudication Parties agree to pursue good-faith negotiations to attempt to resolve certain other claims and any related exceptions that may be filed in the Klamath Adjudication and, subject to subsection 1.5, to use Best Efforts to avoid as much of the litigation in the Klamath Adjudication as possible over such claims. These claims and related exceptions include but need not be limited to:
 - **1.3.1.** The following water rights claims of the Tribes and the BIA in its capacity as trustee for the Tribes and any related exceptions: claims 615 and 623 (Klamath Marsh); claims 614 and 624 (Seeps and Springs); claims 631-633, 635-639 and the relevant portion of claim 612 (Williamson River Basin instream flow claims above Klamath Marsh); claims 663-664 and 665-667 and the relevant portion of claim 612 (Sycan River Basin instream flow claims above Sycan Marsh); and claims 671-673 and the relevant portion of claim 612 (Klamath River instream flow claims below Upper Klamath Lake); and
 - **1.3.2.** The water rights claims of irrigators who are signatories to this Agreement and who, as part of the successful resolution of an individual irrigation claim and any related exceptions also agree, if eligible, to participate in the Riparian Program.
 - 1.3.3. U.S. Fish and Wildlife Service claims 300-320.
- 1.4. Deadline for Filing of Exceptions. The Adjudication Parties do not anticipate that any further requests to extend the deadline for filing exceptions in the Klamath Adjudication in the Klamath County Circuit Court will be made. The Adjudication Parties agree to support, consistent with subsection 12.2.2.1 the current deadline for filing exceptions (May 30, 2014). However, if an Adjudication Party determines that unexpected circumstances warrant an extension, the Adjudication Party shall invoke the Meet and Confer procedures of section 11.

Deferral of Proceedings on Exceptions to Certain Claims. The Adjudication Parties recognize and agree that (1) litigation in the Klamath Adjudication will proceed before the anticipated date of publication of the Affirmative Notice under subsection 10.1 of this Agreement making the Agreement permanent; (2) due to the possibility that, by virtue of action under section 10 of this Agreement, this Agreement may not become permanent the Adjudication Parties must be able to preserve their positions in the Klamath Adjudication; and (3) to preserve their positions in the Klamath Adjudication the Adjudication Parties may have to file in the Klamath County Circuit Court exceptions, responses to exceptions, and/or replies to responses to exceptions concerning the Provisionally Settled Tribal Water Right Claims and/or those additional claims listed in subsection 1.3, and may also participate in litigation that the Adjudication Parties may agree is necessary to determine the right of certain non-Party exceptors to file or litigate exceptions. To minimize litigation in the Klamath Adjudication over the Provisionally Settled Tribal Water Right Claims and any claims listed in subsection 1.3 if the Agreement becomes permanent under the terms of section 10, and for claims listed in subsections 1.3.2 and 1.3.3, and any unresolved claims in 1.3.1, the Adjudication Parties agree to take the following actions:

The Adjudication Parties agree to support through Best Efforts and appropriate means the scheduling of hearings on exceptions filed to the Provisionally Settled Tribal Water Right Claims, and any claims listed in subsection 1.3 that are resolved between the Adjudication Parties, or that are unresolved but as to which the relevant parties to the claim(s) are amenable to further good faith attempts at settlement, so that such hearings or other proceedings on exceptions to such claims do not commence until after either there is sufficient opportunity to explore settlement of the claims listed in subsections 1.3.2 and 1.3.3, and any unresolved claims in 1.3.1, or for the Provisionally Settled Tribal Water Right Claims, there is an Affirmative Notice published under subsection 10.1, or a Negative Notice published under subsection 10.2.

1.6. Conditional Resolution of Off-Reservation Claims. In order to accomplish the conditional resolution of the Off-Reservation Claims as promptly as possible, the BIA, Klamath Tribes, and OWRD agree that within 90 days of the effective date of the Oregon legislation described in subsection 12.4, the BIA, Klamath Tribes, and OWRD will take all necessary steps to secure a partial final decree that accomplishes the following:

- **1.6.1.** Conditionally approves the withdrawal by the Klamath Tribes and the BIA of the Off-Reservation Claims in subsection 1.1., and those Off-Reservation Claims in subsection 1.3.1 that have been subsequently resolved between the Adjudication Parties;
- 1.6.2. Conditionally approves a conditionally amended FFOD in the Klamath Adjudication filed by OWRD with the Klamath County Circuit Court, that (1) acknowledges the conditional withdrawal of the Off-Reservation Claims, (2) conditionally withdraws the findings of fact and conclusions of law pertaining to the merits of the withdrawn Off-Reservation Claims or of the withdrawn parts of partially withdrawn claims, (3) withdraws from the FFOD for Claim 622 (Upper Klamath Lake) the findings of fact and conclusions of law addressing the off-reservation basis for that claim; and (4) any other language in the FFOD addressing the withdrawn Off-Reservation Claims; and
- **1.6.3.** Provides that in the event the Agreement is terminated under section 10 of the Agreement, the Klamath County Circuit Court will enter an order providing that (1) the conditional withdrawal of the Off-Reservation Claims is rescinded; (2) the exceptions to the Off-Reservation Claims are revived; (3) the findings of fact and conclusions of law related to the denial of the Off-Reservation Claims in the FFOD and any other language in the FFOD addressing Federal Indian reserved water rights to off-reservation water are re-entered; and (4) the litigation on the Off-Reservation Claims is resumed.
- **1.7. Adjudication Parties as Agreement Parties.** Individuals or entities that are parties to the Klamath Adjudication that are listed in Exhibit A will become Parties to this Agreement upon their signature. This action is not subject to the provisions of subsection 12.1.3.

2. ECONOMIC DEVELOPMENT

- **2.1. Economic Self-Sufficiency.** The Non-Federal Parties will work to support efforts by the Klamath Tribes to address short-term and long-term economic goals.
- **2.2. Mazama Forest.** The Non-Federal Parties support the acquisition of the Mazama Forest by the Klamath Tribes, and its placement into trust by the United States for the benefit of the Klamath Tribes for economic development and other purposes.
- **2.3. Water Use and Riparian Programs.** The Water Use Program described in section 3, and the Riparian Program described in section 4, together are intended to provide economic and other benefits for out-of-stream water uses, as well as ecosystem benefits that will advance recovery and maintenance of habitats and water quality important to Tribal fisheries and other aspects of the Tribal subsistence economy.
- Economic Self-Sufficiency. The Non-Federal Parties recognize and 2.4. support that the consideration for the settlement of the Provisionally Settled Tribal Water Right Claims and conditional relinquishments of Tribal claims related to certain water resources in section 2.5 is based, in part, on the expectations of the Klamath Tribes and the United States that certain benefits, in addition to those benefits enumerated in the Klamath Basin Restoration Agreement (KBRA) and Klamath Hydroelectric Settlement Agreement (KHSA), will be received by the Klamath Tribes before settlement is complete as described in this subsection 2.4 and in section 10. These benefits are: (1) congressional establishment and subsequent funding of a Tribal economic development fund in the amount of \$40 million, with the parameters of funding, fund access, and use to be established in Federal Authorizing Legislation for the settlement, and (2) the payment of \$1 million annually from the Department of the Interior to the Tribes, for a period of five years, to address Tribal needs during the Transition Period beginning in 2014. The Non-Federal Parties agree to support legislation and administrative actions by the Interior Department so that the requirements of this section are met.
- **2.5. Klamath Tribes' Relinquishment of Claims to the United States and Retention of Rights.** As between the United States and the Klamath Tribes, the relinquishment of claims and retention of rights are limited to the provisions of KBRA Section 15.3.5 and are fully applicable to this Agreement as if set out herein, except with certain adjustments stated in 2.5.1 through 2.5.6 of this Agreement. No additional types of claims

against the United States beyond those specified in the KBRA are relinquished and released under this Agreement.

- **2.5.1**. With regard to the geographic exclusions in KBRA Section 15.3.5.A, if the Tribal claims to water listed in subsection 1.3.1 of this Agreement concerning the area of the Williamson River drainage above Kirk Reef and the Sycan River drainage above the mouth of the Sycan Marsh are subsequently settled, then the provisions of KBRA Section 15.3.5 and subsection 2.5 of this Agreement are fully applicable to those subsequently settled water claims and the areas encompassed within those same water claims.
- 2.5.2. This Agreement is "the OPWAS under Section 16.2.1" envisioned in the KBRA, including Section 15.3.5.A. of the KBRA. The nomenclature used in KBRA Section 15.3.5.A.ii, "contests" and "case in the KBA", references how the Tribal Water Rights Claims were organized in the Klamath Adjudication at that time (the administrative phase of the Adjudication). Currently, the Adjudication is before the Klamath County Circuit Court where the Tribal Water Rights Claims are organized by "claim" and the claims have not yet been organized into "cases"; further, "exceptions" in the court process are the functional equivalent of "contests" in the administrative process. For purposes of the relinquishment of rights under KBRA Section 15.3.5.A.ii, the portions of the Tribal Water Rights Claims that are being resolved in this Agreement are the Provisionally Settled Tribal Water Right Claims.
- **2.5.3.** The promise in KBRA Section 15.3.5.A.iii to relinquish "all claims relating to the negotiation, execution, or adoption of this Agreement" applies to this Agreement as well as to the KBRA and KHSA.
- **2.5.4.** The retention of rights by the Klamath Tribes under KBRA Section 15.3.5.B applies fully to this Agreement, and for purposes of this settlement, the phrase "this Agreement" in that KBRA section covers both the usage of the term in the KBRA and the current usage of "this Agreement" in this document.
- **2.5.5.** The events required to occur before the relinquishment of claims under KBRA Section 15.3.5.A and subsection 2.5 of this Agreement are (1) those enumerated in KBRA Section 15.3.5.C; (2) the events specified in subsection 2.4 of this Agreement, including receipt by the Klamath Tribes of the full payment of the funds (\$45 million total); and (3) those events enumerated in section 10 of this Agreement.

- **2.5.6.** The tolling provisions of KBRA Section 15.3.5.D apply fully to this Agreement and to the claims that are being relinquished.
- **2.6. Jobs Program.** The Non-Federal Parties agree to work with the Klamath Tribes, through the Joint Management Entity (JME), to establish a jobs program for Klamath Tribal members related to implementing this Agreement in the Off-Project Area, as well as to provide opportunities for tribal members to work with individual landowners to enable the Tribal Members to develop the skills and expertise for sustainable ranch management in the Upper Klamath Basin. The goal will be to secure between ten and twenty such positions during the implementation of the actions in the Off-Project Area funded for the period as described in subsection 10.1.6 of this Agreement.

3. WATER USE PROGRAM (WUP)

- 3.1. Overall Administration of the Water Use Program. The Joint Management Entity (JME) is responsible for the design, development and oversight of both the Transitional Water Use Program described in section 5, and the Water Use Program (WUP) described in this section. The Landowner Entity (LE) will be responsible for contacting Eligible Landowners and negotiating, entering into, and administering Water Use Agreements to carry out the Transitional WUP and the WUP, and assuring that the performance requirements set forth in subsections 3.2 through 3.4 of this section are met. The respective roles of the JME, the LE, and the Klamath Tribes are described in more detail in sections 7 and 8 of this Agreement.
- **3.2. Water Use Program Outcomes.** Following the Transition Period described in section 5 of this Agreement, the WUP will achieve two outcomes:
 - **3.2.1.** Permanently increase the total volume of inflow into Upper Klamath Lake over Baseline Conditions by 30,000 acre-feet on an Average Annual Basis (the Total WUP Volume), allocated among WUP Regions, by decreasing the Net Consumptive Use of water as described in subsections 3.3 through 3.8 and 3.12 through 3.15 of this section and in the WUP Guidelines.
 - **3.2.2.** Permanently utilize Call Thresholds and WUP Region Calls as the only bases for calls for water regulation by the BIA and the Klamath Tribes in respect of the water rights that are the subject of the Provisionally Settled Tribal Water Rights Claims during the Irrigation Season. The Call Thresholds may vary depending on hydrologic conditions and the level of compliance that is achieved and maintained in the WUP and the Riparian Program, and may include in certain circumstances the levels of the Tribal Water Rights as specified in subsections 3.9 and 3.10 and Exhibit D.
- 3.3. Increased Inflow into Upper Klamath Lake. Increased inflow into Upper Klamath Lake will come from reductions of Net Consumptive Use of water used for irrigated agriculture in the Off-Project Area, allocated to particular WUP Regions (the WUP Regions are delineated in Exhibit B) as listed below and as presented in Exhibit C. These WUP Region Volumes will be achieved by the end of the Transition Period, as described in section 5, and then maintained permanently. The WUP Region Volumes, along with the Call Thresholds described in subsection 3.9 and 3.10, are among the expected benefits of the WUP to the Klamath Tribes and the United States. Those WUP Volumes are:

- 3.3.1. Upper Sprague WUP Region Volume: 6,900 acre-feet;
- **3.3.2.** Sycan WUP Region Volume: 1,050 acre-feet;
- **3.3.3**. Lower Sprague WUP Region Volume: 9,420 acre-feet;
- **3.3.4.** Middle Williamson WUP Region Volume: 330 acre-feet;
- **3.3.5.** Lower Williamson WUP Region Volume (including Modoc Point Irrigation District): 2,700 acre-feet; and.
- 3.3.6. Wood Valley WUP Region Volume: 9,600 acre-feet.
- **3.4. Other Water Management Changes.** Water management changes implemented outside of the Off-Project Area may result in increased inflow into Upper Klamath Lake; however, any resulting increases in inflow will not be counted toward the achievement of WUP Region Volumes.
- 3.5. WUP Guidelines. The WUP Guidelines will be developed by the JME Technical Team, based on the documents prepared by the preagreement technical team and attached as Exhibits E and K. The WUP Guidelines and any subsequent amendments to the WUP Guidelines must be approved by the JME, and by the United States Fish and Wildlife Service (USFWS) if the United States is not already a voting member of the JME. In the event of a conflict between the preceding documents and this Agreement, this Agreement will control. The JME will approve and amend the WUP Guidelines as a Major Decision. The WUP Guidelines will include at a minimum:
 - **3.5.1.** Methods for estimating the changes in Net Consumptive Use resulting from implementation of WUP Practices;
 - **3.5.2.** Methods for estimating how changes in Net Consumptive Use will change flow volumes within the applicable WUP Region; and
 - **3.5.3.** Eligible WUP Practices, including those WUP Practices listed in subsection 3.12, and others subsequently approved by the JME, for which supporting analyses provide credible evidence that Net Consumptive Use will be decreased and instream flows increased as a result of their implementation.
 - **3.5.4.** Procedures for estimating the change in Net Consumptive Use expected to result from implementing a WUP Practice on a specific

- parcel. The change in Net Consumptive Use is computed as the difference between the Initial Net Consumptive Use and the Post-Implementation Net Consumptive Use.
- **3.5.5.** Procedures for estimating the change in instream flow volumes described in section 3.5.2 to be listed in the WUP Ledger as debits or credits consistent with subsections 3.7 and 3.8. The procedures will follow the logic and calculations embodied in the applicable Excel spreadsheets developed by the Parties in support of this Agreement to estimate such instream flow changes. The spreadsheet named Wood Valley OPWAS Final 1Mar2014.xlsm will be used for those portions of the Off-Project Area located in the Wood Valley. The spreadsheet named Williamson OPWAS Final 1Mar2014.xlsx will be used for those portions of the Off-Project Area located in the Williamson, Sprague, and Sycan river drainages. These Excel files have been reviewed by the Parties and will be provided to the JME and LE in electronic form.
- **3.5.6.** Procedures for monitoring and reporting the status of WUP Practices by the Eligible Landowner to the LE, and by the LE to the JME, as appropriate.
- **3.5.7.** Procedures for prioritizing selection of parcels that have been offered by willing Eligible Landowners for enrollment in Water Use Agreements or for implementing WUP Practices. The procedures shall consider the following factors, or other factors that may subsequently be determined by the JME. The Parties acknowledge that such prioritizations may be complex, and at times may require the JME to address difficult trade-offs among conflicting priorities. The priorities will include, but are not limited to:
- **3.5.7.1.** Increased priority for WUP Practices on lands that have a higher than average Net Consumptive Use in order to minimize the acreage of lands affected, make the most effective use of limited funding, maximize instream flow benefits, or minimize the proportion of WUP Region Volumes that must be accomplished by means other than retirement, transfer, or lease of water rights;
- **3.5.7.2.** Increased priority for parcels irrigated by surface water that are farther from a perennial stream than other parcels;
- **3.5.7.3.** Increased priority for parcels irrigated by surface water that are higher in elevation relative to the nearest perennial stream;
- **3.5.7.4.** Increased priority for parcels irrigated by groundwater that are nearer to a spring or to a Gaining Reach of a perennial stream,

with appropriate consideration for the depth of the well, characteristics of well construction, groundwater simulation results, and other pertinent information;

- **3.5.7.5.** Increased priority for water rights with priority dates after 1905;
- **3.5.7.6.** Increased priority for parcels on which reduced Net Consumptive Use will increase instream flows in streams or reaches of particular biological significance;
- **3.5.7.7**. Increased priority for parcels on which reduced Net Consumptive Use will increase instream flows in streams to a greater extent on a per acre basis; and
- **3.5.7.8.** Increased priority for parcels on which higher amounts of water are projected to be realized instream relative to the cost.

3.6. Water Use Agreements.

- **3.6.1.** The LE will enter into Water Use Agreements with Eligible Landowners to reduce the Net Consumptive Use of water. In addition, the LE will document Water Use Development and other WUP Practices that are not contained in a Water Use Agreement. Each Water Use Agreement must be reviewed by the JME Technical Team, which will evaluate the technical adequacy of the proposed WUP Practices, as well as the extent to which the WUP Guidelines were followed. The JME Technical Team will make recommendations to the JME on these issues. Then the JME will approve the WUP Practices if the supporting information is adequate and the WUP Guidelines have been followed. Before becoming final, each Water Use Agreement must be approved by the JME as a Major Decision, as well as by the LE and the Eligible Landowner. The JME must act within 30 days to either approve or disapprove the Water Use Agreement, and to approve, modify or disapprove the associated WUP Ledger entry. The Parties acknowledge that collaborative relationships among the Klamath Tribes, the United States, the LE, and Eligible Landowners will help facilitate the rapid approval of Water Use Agreements by the JME.
- **3.6.2.** Each Water Use Agreement will identify how the obligations in the agreement may be enforced. In general, each Water Use Agreement will provide that in the event of a failure to perform the Eligible Landowner's obligations, the LE will first give notice to the Eligible Landowner of the default. If the landowner fails to cure the default, within the time period specified in the Water Use Agreement, the LE

must give notice to the JME and the Klamath Tribes, and may enforce the obligation through specific enforcement of the Water Use Agreement. If the default continues for a period of more than 60 days after notice to the Eligible Landowner, each Water Use Agreement will provide that the Klamath Tribes and the United States each have the right to enforce the agreement, and the right to obtain reimbursement for their expenses.

- **3.7. Tracking WUP Region Volumes.** The Parties will use a WUP Ledger to track the WUP Region Volumes over time to determine ongoing compliance. The LE will designate a staff person to make draft entries into the WUP Ledger, according to the sections 3.7.1 through 3.7.4 and the WUP Guidelines.
 - **3.7.1.** A final entry in the WUP Ledger will be made as a Major Decision when a Water Use Agreement is approved by the JME, or Water Use Development or another WUP Practice is approved by the JME. When a Water Use Agreement or another WUP Practice is proposed, the LE will estimate the value of the associated entry into the WUP Ledger in accordance with the following steps:

Step One: The value of the entry will be the estimated average annual change to instream flow volume within the WUP Region resulting from the Water Use Agreement or other WUP Practice or Water Use Development determined as follows and as in subsections 3.7.2 and 3.7.3.

Step Two: Estimated increases in instream flow volume resulting from implementing WUP Practices that reduce Net Consumptive Use will be credited towards the applicable WUP Region Volume in the WUP Ledger based on the difference between Initial Net Consumptive Use and the Post-Implementation Net Consumptive Use.

Step Three: Estimated decreases in instream flow volume resulting from increases in Net Consumptive Use due to the termination, change, or reduction of WUP Practices or due to Water Use Development will be debited in the WUP Ledger.

Step Four: The JME will assure that WUP Ledger entries are consistent with this section as well as the WUP Guidelines based on the recommendations of the JME Technical Team and that actions and practices identified in Sections 18.2.1 through 18.2.3 of the KBRA are not counted in the WUP Ledger. Only WUP Practices and Water Use Development undertaken after December 31, 2001 and still in effect will be included in the WUP Ledger.

Step Five: Each entry in the WUP Ledger will be indexed to the information developed or relied upon to justify the entry.

- **3.7.2.** The Initial Net Consumptive Use for a parcel where a WUP Practice or Water Right Development is occurring or is proposed will be determined based on the evapotranspiration analysis reported in the U.S. Geological Survey (USGS) Open-File Report No. 2012-1199, entitled Hydrological Information Products for the Off-Project Water Program of the Klamath Basin Restoration Agreement. For any parcel, the average of the Net Consumptive Use estimates from this analysis for the years 2004 and 2006 will comprise the estimate for the Initial Net Consumptive Use, unless more accurate estimates are subsequently developed and agreed on as a Major Decision of the JME. Estimates of Post-Implementation Net Consumptive Use will also be based on the USGS Report, pending subsequent work to improve the accuracy of Net Consumptive Use estimates. The Parties agree to support such work. The new information may rely on new methods and/or may result from applying the methods reported in the USGS Report to more years before and after implementation of the WUP. Improving the accuracy of Post-Implementation Net Consumptive Use estimates may take multiple years of measurements as vegetation shifts from irrigated land to nonirrigated land in areas where water rights are permanently retired. If the JME determines that new information is more accurate for purposes of determining Initial or Post-Implementation Net Consumptive Use, the JME may alter the estimates used in the WUP Guidelines as a Major Decision. If such a change is made, it will not alter the contractual aspects of the Water Use Agreements that have already been executed, but will be incorporated into subsequent Water Use Agreements. Such changes may result in amendments to the WUP Ledger, but such amendments will not be deemed to cause a shortfall from WUP Region Volume obligations that would trigger a WUP Region Call or a Type A SIF Adjustment until and unless the shortfall remains three months after the amendment to the WUP Ledger.
- **3.7.3.** The change in Net Consumptive Use resulting from a WUP Practice will be determined by subtracting the parcel-specific estimate of Post-Implementation Net Consumptive Use from the estimate of Initial Net Consumptive Use, and multiplying this difference by the number of acres affected by the WUP Practice. If the WUP Practice applies for only a portion of the Irrigation Season, the estimate of the Post-Implementation Net Consumptive Use will account for the monthly variations associated with that WUP Practice.

3.7.4. The JME Technical Team will review the supporting information for each proposed WUP Ledger entry, and make recommendations to the JME as to the adequacy of the supporting information and whether the WUP Guidelines were followed. Once approved by the JME, a WUP Ledger entry will be confirmed and included in the WUP Ledger. When a confirmed entry is made in the WUP Ledger, the cumulative total volume will be updated. If the cumulative total volume is below the applicable WUP Region Volume, then the LE will take immediate action to address the shortfall, and the Klamath Tribes and the BIA may make a call for regulation of water use as provided in subsection 3.8 to make up the shortfall.

3.8. Use of the WUP Ledger.

- **3.8.1.** The WUP Ledger is the basis for determining compliance with the WUP Region Volumes and making WUP Region Volume calls, as described in this subsection 3.8.
- **3.8.2.** If there is a shortfall in a WUP Region Volume, the Klamath Tribes and the BIA may call for regulation for the Tribal Water Rights as described in this subsection 3.8 to the extent necessary to make up the shortfall.
- **3.8.3.** If the Klamath Tribes and the BIA make a regulatory call under this subsection 3.8, it will first be to the amount of the applicable Tribal Water Right within the applicable WUP Region. If that call is not sufficient to address the shortfall, then the call will next be made to the Tribal Water Right(s) in the nearest downstream claim reach or reaches, to the extent necessary to eliminate the shortfall. If those call(s) are insufficient to address the shortfall, then the Klamath Tribes and the BIA may make a call to the Tribal Water Right for lake elevations in Upper Klamath Lake, to the extent of the shortfall. If a call from Upper Klamath Lake will not generate additional instream flow in the non-compliant WUP Region, then the call for the Tribal Water Right in Upper Klamath Lake will end once any shortfall in the Total WUP Volume is eliminated. Once a shortfall in the WUP Region Volume is corrected, then the applicable calls for regulation by the Klamath Tribes and the BIA will return to the applicable Call Threshold(s).
- **3.9. Specified Instream Flow (SIF) Call Thresholds.** The WUP is designed to increase instream flow rates in the Off-Project Area. Within the Off-Project Restoration Area, these increased flows are expected to meet or exceed SIF Call Thresholds in most years. A Call Threshold is the instream flow rate threshold associated with a Primary or Secondary

SIF Measurement Location, from which the Klamath Tribes and the BIA may call for regulation of junior water rights under the terms of subsections 3.9, 3.10 and 3.11.

- **3.9.1.** Each SIF Measurement Location shown in Exhibit B is associated with a Long-Term SIF Threshold. When sufficient data are available, a Long-Term SIF Threshold is based on historic stream flow gage data, adjusted to reflect the instream flow results of restoring the WUP Region Volumes instream. In such cases, when conditions are relatively wet, Long-Term SIF Thresholds will be at or near the applicable Tribal Water Right for that SIF Region, but never higher (but see subsection 3.7 and 3.8. regarding shortfalls in WUP Region Volumes). As conditions become drier, the Long-Term SIF Thresholds decline, eventually reaching absolute minimum values under extremely dry conditions. At those locations where long-term data are insufficient to support such an approach, Long-Term SIF Thresholds are established by allocating a portion of the water supply to instream flows. The Long-Term SIF Thresholds are shown in Exhibit D.
- **3.9.2.** There are two types of SIF Measurement Locations: Primary and Secondary, as shown on Exhibit B.
- **3.9.2.1.** If a stream flow falls below the applicable Call Threshold at a Primary SIF Measurement Location, the Klamath Tribes and the BIA may call for regulation of junior water rights as necessary to meet the Call Threshold.
- **3.9.2.2.** If stream flows fall below the applicable Call Threshold at a Secondary SIF Measurement Location, the LE will work with Eligible Landowners to correct the shortfall. If the shortfall continues for a period exceeding five days, then the Klamath Tribes and the BIA may call for regulation of junior water rights as necessary to meet the Call Threshold.
- **3.9.3.** Call Thresholds at Primary and Secondary SIF Locations are calculated for each month during the Irrigation Season. For example, on April 1, hydrologic conditions experienced in March are used to predict the magnitude of low flows during the coming April, which establishes the expected relative wetness or dryness of April. Once this predicted value is known, it is used to identify the Long-Term SIF Threshold that coincides with these predicted hydrologic conditions. This process is repeated for each month of the Irrigation Season. If conditions within the applicable SIF and WUP regions are fully compliant with the terms of this Agreement, then the Call Threshold will be the same as the Long-Term SIF Threshold. Otherwise, the Call

Threshold will be incrementally adjusted upwards until compliance is established, or re-established, as described in subsections 3.9 and 3.10, at which time the Call Threshold will return to the Long-term SIF Threshold. Specifically, a Call Threshold = Long-Term SIF Threshold + Type A SIF Adjustment + Type B SIF Adjustment. SIF adjustments are described in subsections 3.9 and 3.10. Detailed methods for calculating Call Thresholds are described in Exhibits D and E.

- **3.9.4.** If a Call Threshold at a Primary SIF Measurement Location is not met, or if a Call Threshold at a Secondary SIF Measurement Location is not met for a period exceeding five days, then calls to the applicable Call Threshold may occur consistent with subsections 3.9 and 3.10. In addition, if a Call Threshold is not met, then the applicable Call Threshold for the next month is adjusted upwards, by the amount of the Type A SIF Adjustment, as described in the following subsections and in more detail in Exhibit E.
- **3.9.4.1.** On the first day of each month of the Irrigation Season, the average percentage shortfall (if any) from the Long-Term SIF Threshold for the five days with lowest average daily flows during the preceding month will be determined. October and March are considered to be consecutive for the purposes of this subsection.
- **3.9.4.2.** Starting in the second month of a continuing shortfall at a SIF Measurement Location, one percent will be added to the percentage shortfall determined under subsection 3.9.4.1. For each month that the shortfall continues thereafter, another one percent will be added, so that, for example, in the fourth month of a continuing shortfall, three percent will be added to the percentage shortfall determined under subsection 3.9.4.1, as a duration component adjustment. October and March are considered to be consecutive for the purposes of this subsection. If there is an Extreme Drought then the duration component of the Type A SIF Adjustment under this subsection 3.9.4.2 will not be included in the calculation of the Type A SIF Adjustment for the duration of the Extreme Drought. At the end of the Extreme Drought declaration, the calculation will revert to the duration component adjustment at the time the Extreme Drought was declared.
- **3.9.4.3.** The percentage determined in subsections 3.9.4.1 and 3.9.4.2 will then be multiplied by the Long-Term SIF Threshold to determine the Type A SIF Adjustment as described in Exhibit E. That amount will be added to the Long-Term SIF Threshold as part of the calculation of the Call Threshold for the month. The Type A SIF

Adjustment increases the Call Threshold by an amount that is commensurate with the magnitude and duration of the shortfall.

- **3.9.5.** Once a Call Threshold is met for a calendar month, the Type A SIF Adjustment will reset to zero.
- **3.9.6.** Notwithstanding any other provision of this Agreement, in no event will the BIA or the Klamath Tribes have the right under this Agreement to call for regulation of junior water rights from a SIF Measurement Location during the Irrigation Season in the Off-Project Restoration Area beyond those amounts established in the Tribal Water Right(s) applicable to that SIF Measurement Location. Nothing in this Agreement enlarges the Tribal Water Rights above the amount specified in the FFOD. Similarly, notwithstanding any other provision of this Agreement, in no event will the level at which the Klamath Tribes and the BIA may call for regulation of junior water rights from SIF Measurement Locations during the Irrigation Season in the Off-Project Restoration Area fall below the minimum flow levels shown in Exhibit D.
- **3.9.7.** This Agreement does not affect Tribal Water Rights outside of the Irrigation Season.
- 3.10. Regulatory Calls Resulting from Nonperformance Under the Riparian Program. As described in subsections 3.2.2, 3.9, and 3.10, under this Agreement, water rights within the Off-Project Restoration Area are subject to regulation based on Call Thresholds at a network of SIF Measurement Locations. Applicable Call Thresholds will be subject to upward adjustment by the Type B SIF Adjustment as described in this subsection 3.10, subsection 4.12, and Exhibit E if the terms of the Riparian Program are not met. Calculation of the Type B SIF Adjustment will take one of two tracks in 3.10.1 or 3.10.3.
 - **3.10.1**. If implementation of the Riparian Program does not achieve Sufficient Participation by Eligible Riparian Landowners within a SIF Region by the end of the Transition Period, then the Type B SIF Adjustment will be calculated as described below.
 - **3.10.1.1.** At the beginning of each month during the Irrigation Season, it will be determined whether Sufficient Participation has been achieved in each SIF Region, as provided in section 4. If Sufficient Participation has not been achieved, the percent of shortfall for that month will be determined as specified in subsections 4.8, 4.9, 4.12, 5.4 5.5, 7.2.10, 8.3, and Exhibit E. A shortfall could result from an inadequate geographic extent of Riparian Management Agreements,

Eligible Riparian Landowners that are out of compliance with the terms of their Riparian Management Agreement, or both.

- **3.10.1.2.** If a shortfall from Sufficient Participation persists for two or more consecutive months, then starting in the second month and continuing in all subsequent months of the Irrigation Season, the number of consecutive months of shortfall will be summed, with each month subsequent to the first month of the shortfall counting as one percent. For the purposes of this subsection, October and March are considered to be consecutive. The percentages will be added, and then multiplied by the Long-Term SIF Threshold to produce the Type B SIF Adjustment for the coming month, as described in more detail in Exhibit E.
- **3.10.1.3.** Once compliance is achieved, the Type B SIF Adjustment will be reset to zero.
- **3.10.2.** If an Eligible Riparian Landowner chooses to retire all irrigation water rights that are subject to a call from Tribal Water Rights on a parcel and that parcel includes land within the Riparian Management Corridor, then that Eligible Riparian Landowner must execute a Riparian Management Agreement for the Riparian Management Corridor within that parcel. After Sufficient Participation has been reached within a SIF Region, the linear extent of such Riparian Management Agreements will be called the "Retired Riparian Length," up to a maximum of 40% of the length required for Sufficient Participation. The Retired Riparian Length and compliance with Riparian Management Agreements will be tracked in the RMA Ledger described in subsection 4.12.
- **3.10.3.** After achievement of Sufficient Participation by Eligible Riparian Landowners within a SIF Region, the Type B SIF Adjustment will be calculated as follows. The Retired Riparian Length (up to 40% of the length required for Sufficient Participation) and the length of other lands within the applicable Riparian Management Corridor which are in compliance with an applicable Riparian Management Agreement, as shown in the RMA Ledger, shall both be subtracted from the Sufficient Participation length required for that SIF Region. This difference shall be divided by the Sufficient Participation length. This quotient is the percentage shortfall for the applicable month. If a shortfall persists for two or more consecutive months, then starting in the second month and continuing in all subsequent months of the Irrigation Season the number of consecutive months of shortfall will be summed, with each month subsequent to the first month of the shortfall counting as one percent. For the purposes of this subsection, October and March are

considered to be consecutive. The percentages will be added, and then multiplied by the Long-Term SIF Threshold to produce the Type B SIF Adjustment for the coming month, as described in more detail in Exhibit E. Once Sufficient Participation is met, the Type B SIF Adjustment will be reset to zero.

3.10.4. If a shortfall from Sufficient Participation within a SIF Region triggers the procedures in subsections 3.10.1 – 3.10.3, and no increase in the Call Threshold at the SIF Measurement Location is possible because it is already equal to the Tribal Water Right associated with that SIF Measurement Location, then the Type B SIF Adjustment will be treated as if it were a shortfall from a WUP Region Volume, and regulation may proceed following subsection 3.8.2. If such regulation pursuant to subsection 3.8.2 results in a call from Upper Klamath Lake that will not generate additional instream flow in the non-compliant SIF Region, then the call will not proceed.

3.11. Groundwater Regulation.

- **3.11.1.** An Eligible Landowner may hold both surface water rights and groundwater rights for use on the same lands. In these situations, under Oregon law, when a primary surface water right is transferred or leased instream, the use of supplemental groundwater is prohibited on the lands from which the surface water right is transferred. OAR 690-380-2250.
- **3.11.2.** The Parties agree that nothing in this Agreement provides any additional limitation on the use of groundwater pursuant to ORS 537.545.
- **3.11.3.** The Parties, other than the OWRD, agree that regulation of groundwater with a point of appropriation in the Off-Project Area that is less than one-quarter mile from a Gaining Reach should be consistent with OAR Chapter 690 Division 009, as described in subsections 3.11.3.1 and 3.11.3.2, below.
- **3.11.3.1.** The Parties, other than the OWRD, agree that the use of groundwater with a point of appropriation in the Off-Project Area that is no more than 500 feet from a Gaining Reach may be regulated in favor of senior surface water rights.
- **3.11.3.2.** The Parties, other than the OWRD, agree that the use of groundwater with a point of appropriation in the Off-Project Area that is greater than 500 feet and less than one-quarter mile from a

Gaining Reach may be regulated in favor of senior surface water rights if regulation would provide relief in an effective and timely manner.

The Parties acknowledge that the best information presently available to the OWRD supports a presumption that regulation of such groundwater rights will provide relief in an effective and timely manner, pursuant to OAR 690-009-0050(2)(a). The OWRD is currently performing well-specific modeling for groundwater sources in this category. The OWRD will provide the results of this modeling to the Parties, and these results are expected to provide more accurate information than is currently available and will be used to determine if the presumption is valid. In addition, the water right holder or other interested parties may provide their own evidence to the OWRD pertaining to the validity of the presumption that a point of appropriation in the Off-Project Area that is located greater than 500 feet and less than one-quarter mile from a Gaining Reach will provide relief to senior surface water right holders in an effective and timely manner. If a person disagrees with the enforcement determination of the water master they may seek recourse per Applicable Law.

- **3.11.4.** The Parties, other than the OWRD, agree that, except as provided in subsection 3.11.5, the use of groundwater with a point of appropriation in the Off-Project Area that is between one-quarter mile and one mile of a Gaining Reach should only be regulated to satisfy senior surface water rights or Call Thresholds (as applicable), as described in subsections 3.11.4.1 to 3.11.4.5, below.
- **3.11.4.1.** The Parties, other than the OWRD, agree that, except as provided in subsection 3.11.4.4, if a valid call is made by a senior surface water right holder, and the OWRD determines that the difference between the rate (i.e., the amount expressed in cfs) of the shortfall of water validly called and the rate of the senior water right or Call Threshold (as applicable) is between 5% and 10% of the amount of the senior water right call or the Call Threshold (as applicable), then the use of groundwater with a point of appropriation between onequarter mile and one-half mile of a Gaining Reach may be regulated. However, the Parties, other than the OWRD also agree that if the first such valid call based on a specific senior water right or Call Threshold (as applicable) is made after August 31, then such groundwater sources should not be regulated during that Irrigation Season, as such regulation likely will not provide effective and timely relief. For example, if a senior user makes a valid call based on a water right or Call Threshold, as applicable, of 100 cfs, and the Watermaster determines the flow (measured at the appropriate location) is 93 cfs, then the shortfall is 7 cfs. This equates to a 7% shortfall, which under

this provision has the result that wells out to one-half mile may be regulated to satisfy the call.

- **3.11.4.2.** The Parties, other than the OWRD, agree that, except as provided in subsection 3.11.4.4, if a valid call is made by a senior surface water right holder, and the OWRD determines that the difference between the rate (*i.e.*, the amount expressed in cfs) of the shortfall of water validly called and the rate of the senior water right or Call Threshold (as applicable) is greater than 10% of the amount of the senior water right call or the Call Threshold (as applicable), then the use of groundwater with a point of appropriation greater than one-half and up to one mile from a Gaining Reach may be regulated. However, the Parties, other than the OWRD, also agree that if the first such valid call based on a specific senior water right or Call Threshold (as applicable) is made after July 31, then such groundwater sources should not be regulated during that Irrigation Season, as such regulation likely will not provide effective and timely relief.
- **3.11.4.3.** The Parties, other than the OWRD, agree that if a valid call is made by a senior surface water right holder, and the OWRD determines that the difference between the rate (*i.e.*, the amount expressed in cfs) of the shortfall of water validly called and the rate of the senior water right or Call Threshold has been greater than 5% for a specific senior water right or Call Threshold (as applicable) for more than thirty-one days within a forty-five day period, the groundwater sources with a point of appropriation between one-quarter mile and one mile of a Gaining Reach may be regulated.
- **3.11.4.4.** The Parties, other than the OWRD, agree that if a call is made to a Call Threshold after the 25th day of the month, then groundwater sources with a point of appropriation between one-quarter mile and one mile of a Gaining Reach should not be regulated, as such regulation likely will not provide effective and timely relief, unless the difference between the rate (*i.e.*, the amount expressed in cfs) of the shortfall of water validly called and the rate of the Call Threshold is greater than 10% of the amount of the senior water right or the Call Threshold (as applicable).
- **3.11.4.5.** The Parties, other than the OWRD, agree that, for the purposes of subsections 3.11.4.1. to 3.11.4.3., groundwater sources with a point of appropriation that is located between one-quarter and one-half mile of a Gaining Reach, but that are continuously cased and continuously sealed to a minimum depth of 500 feet below land surface, should be regulated as if they are located between one-half and one mile of a Gaining Reach. In addition, the

Parties, other than the OWRD, agree that groundwater sources with a point of appropriation located greater than one-half mile from a Gaining Reach, but continuously cased and continuously sealed to a minimum depth of 500 feet below land surface, should be regulated as if they are located greater than one mile from a Gaining Reach, and should not be subject to regulation in the absence of a critical groundwater determination.

- **3.11.5.** The Parties, other than the OWRD, agree that notwithstanding subsections 3.11.3 and 3.11.4, a groundwater source in the Off-Project Area with a point of appropriation within one mile of a spring or stream should be regulated as provided in OAR Chapter 690 Division 009 if use of the groundwater source would result in depletion of the flow of a Gaining Reach at a rate greater than 25 percent of the rate of appropriation within 30 days of pumping. This determination will be based on the best available information, which could include employing at least one of the methods set forth in OAR 690-009-0040(4)(d), as provided in OAR 690-009-0050(2)(a). Prior to making such a determination, the OWRD will notify the water right holder subject to the call and the party or parties making the call, and provide them with an opportunity to submit evidence to the OWRD. If a person disagrees with the enforcement determination of the water master they may seek recourse available under Applicable Law.
- **3.11.6.** The Parties, other than the OWRD, agree that notwithstanding subsections 3.11.3. and 3.11.4, groundwater sources in the Off-Project Area with points of appropriation located within a one-mile radius of a particular spring in the Off-Project Area should be regulated consistent with OAR Chapter 690 Division 009 if use of all such groundwater sources would result in depletion of the spring flow rate in an amount that is greater than 20 percent within 30 days. This determination will be based on the best available information, which could include employing at least one of the methods set forth in OAR 690-009-0040(4)(d), as provided in OAR 690-009-0050(2)(a). Prior to making such a determination, the OWRD will notify the water right holder subject to the call and the party or parties making the call, and provide them with an opportunity to submit evidence to the OWRD. If a person disagrees with the enforcement determination of the water master they may seek recourse available under Applicable Law.
- **3.11.7.** The Parties, other than the OWRD, agree that for purposes of subsections 3.11.3 to 3.11.6, distances from individual wells to springs, streams, or Gaining Reaches, as applicable, will initially be determined based on the location of individual wells as shown in Exhibit F, and the location of the spring or the edge of the water visible in the National

Agricultural Inventory Program (NAIP) imagery for July 15 – August 1, 2012, subject to the provisions regarding such distances in subsections 3.11.7.1 to 3.11.7.6, below. The LE will work with OWRD to confirm the accuracy of well locations in Exhibit F; any errors will be corrected and the revised locations will be used for measuring distances. For the purposes of measuring distances from individual wells to springs, streams, or Gaining Reaches, as applicable, resulting from the changes described in subsections 3.11.7.1 through 3.11.7.5, the Parties, other than OWRD, agree to use the most current year of NAIP imagery, or other remote sensing product designated for this use by the JME.

- **3.11.7.1.** If a replacement or additional well under an existing registration, permit, or certificate is located at a distance greater than one mile from a surface water source, pursuant to OAR Chapter 690 Division 009 the well may not be regulated without a critical groundwater area determination.
- **3.11.7.2.** If a riparian restoration action results in movement of the edge of a surface water body in the Off-Project Area to an extent that would change the exposure of a groundwater point of appropriation to regulation based on the distance measurement criteria in subsections 3.11.1 to 3.11.6, then for purposes of subsections 3.11.3 to 3.11.6, the distance prior to the restoration action shall continue to apply.
- **3.11.7.3.** A replacement or additional well under an existing registration, permit or certificate shall be evaluated for the purposes of subsections 3.11.3 through 3.11.6 based on the distance criterion applicable to the original well; except that for the purpose of determining effective and timely relief, the replacement or additional well's measured distance, according to the applicable criterion, shall be used.
- **3.11.7.4.** The JME may recommend to OWRD, based on evidence submitted by a Party or Parties, whether a natural change in stream location has caused a material change in the distance of a well to a Gaining Reach or stream for purposes of subsections 3.11.3 through 3.11.5. If OWRD determines that there is a material change, it will notify the JME and affected person(s).
- **3.11.7.5.** If the best available information regarding the location of a Gaining Reach changes, the JME may recommend to OWRD that Exhibit F be amended, and OWRD shall notify the JME if it proposes to make such a change.

- **3.11.7.6.** If a stream channel moves as a result of either natural or human activities, then before the JME makes a recommendation in subsections 3.11.7.4 or 3.11.7.5 it will seek advice from the JME Technical Team regarding the potential need to allow the stream channel to stabilize in its new configuration before making such recommendations.
- **3.11.8.** The OWRD agrees to propose to the Oregon Water Resources Commission that the Commission adopt, pursuant to OAR 690-009-0030, local rules applicable to groundwater uses with a place of use within the Off-Project Area that are consistent with subsections 3.11.3 through 3.11.7 of this section within 60 days after the Effective Date, and to recommend adoption of such rules. The proposed rules will provide that they apply to calls for regulation that affect water rights in the Off-Project Area, regardless of where the call is made from. If the proposed rule is consistent with this section, the Parties, other than OWRD, agree to support the adoption by the Commission of the proposed rules. In the event that the Commission fails to adopt the proposed rules, the Parties other than OWRD agree that groundwater regulation will be in accordance with OAR Chapter 690 Division 009, as of December 2, 2013, notwithstanding anything in subsections 3.11.3 through 3.11.7 to the contrary.
- **3.11.9.** The Parties agree that, pursuant to OAR Chapter 690 Division 009, wells greater than one mile from a surface water source may not be regulated without a critical groundwater area determination.
- **3.11.10.** A groundwater right with a point of appropriation less than one quarter mile from a Gaining Reach is eligible for retirement and compensation under the WUP. The LE may petition the JME to include other groundwater rights as eligible for retirement and compensation if, upon recommendation from the JME Technical Team, the JME finds that such retirement would have particular value to the Water Use or Riparian Programs; such finding will be a Major Decision. The JME will set aside a portion of funding for each WUP Region to be used to retire groundwater rights, with such retirements taking into consideration the frequency and future risk of regulation as well as the instream flow or riparian benefits.
- **3.11.11.** After the Effective Date of this Agreement, if within a WUP Region the number of additional wells exempted from the requirement to obtain a permit under ORS 537.545 that are registered within the Off-Project Area increases by more than 25 percent over the amount existing as of the Effective Date, then the Parties will Meet and Confer to determine whether additional limitations on groundwater use within

the Off-Project Area should be proposed, or other remedy implemented, in order to ensure that benefits bargained for in this Agreement continue to be provided. If it becomes necessary to regulate the use or distribution of groundwater wells that are exempt under ORS 537.545, the Parties recognize that the regulation shall be by priority date as indicated by the well log filed with the OWRD for that particular exempt use.

- **3.11.12.** The Parties agree that if OWRD permits the development of new irrigation wells in the Off-Project Area, then any such new irrigation wells that are developed in the Off-Project Area will be considered to be Water Use Development and will be debited in the WUP Ledger as specified in subsection 3.7.
- 3.12. Means of Achieving WUP Volumes and Call Thresholds. The primary means of providing the WUP Region Volumes and meeting Call Thresholds will be the reduction of Net Consumptive Use through the permanent retirement of water rights carried out through Water Use Agreements. In addition, however, the WUP may include other WUP Practices that reduce Net Consumptive Use and help achieve WUP Region Volumes and/or meet Call Thresholds or otherwise contribute to instream flows in a quantifiable, predictable manner. These may include, but are not limited to:
 - **3.12.1.** Long-term and short-term leasing of water rights for instream use, including split season leasing;
 - **3.12.2.** Water conservation and efficiency measures that reduce the Net Consumptive Use of water;
 - **3.12.3.** Agreements to forbear the use of water right claims in the Klamath Adjudication;
 - **3.12.4.** Agreements to rotate the use of water among water right holders;
 - **3.12.5.** Storage (natural or artificial) and release of water in order to meet one or more Call Thresholds during periods of low stream flows; and
 - **3.12.6.** Land or water management in uplands including juniper removal, crop rotations, improved soil condition and management, and other similar measures.

- 3.13. Monitoring of Alternate Methods. To the extent that WUP Practices other than the permanent retirement of water rights are used to achieve WUP Region Volumes, the JME, consistent with the recommendations of its Technical Team, will quantify the change in Net Consumptive Use resulting from such WUP Practices. The LE will monitor those WUP Practices to ensure that the reductions are achieved and continued in accordance with the WUP Guidelines. If reductions in Net Consumptive Use are not being achieved or continued, the JME will work with the LE to institute changes or shifts in WUP Practices to achieve the WUP Region Volumes. The JME will work with the LE, OWRD, USGS, and Funding Entities to develop appropriate water metering, measuring, and reporting systems to quantify and monitor the results of WUP Practices.
- **3.14. Water Use Agreements.** The WUP will be implemented through Water Use Agreements entered into on a willing basis between Eligible Landowners and the LE to achieve the outcomes described in section 3. If an Eligible Landowner submits a statement of intent pursuant to subsection 3.22, the LE with the Eligible Landowner will develop a draft Water Use Agreement based on the WUP Guidelines, and a template form of Water Use Agreement approved by the JME. Water Use Agreements must be approved by the JME as a Major Decision.
- 3.15. Willing Sellers. Landowners permanently retiring water rights under the WUP will be compensated based upon values mutually agreed to by purchaser and seller. The Parties recognize that the value of water rights in the Off-Project Area is in a state of uncertainty and flux as a result of the FFOD and the ongoing Klamath Adjudication. In recognition of the uncertainty and the intent to resolve further litigation, the Parties agree the compensation value for retirements will be based on the value of the water rights on March 1, 2013. The Parties also anticipate that retirement of water rights and other means of reducing water use may have a higher value in certain locations. Eminent domain will not be used to acquire water rights under the WUP.
- **3.16. Water Banking.** The JME and the LE may develop a short-term water use reduction program, which may include but is not limited to water right leasing to address atypical conditions when Call Thresholds are not met.
- 3.17. Relationship Between Water Use Agreements and Riparian
 Management Agreements. An Eligible Landowner who enters into a
 Water Use Agreement must also enter into a Riparian Management

Agreement under section 4 if the parcel subject to the Water Use Agreement includes land within a Riparian Management Corridor.

- **3.18. Limitations on the Retirement of Water Rights.** The WUP will not exceed the following limitations on the total number of acres that may be permanently retired, unless the LE is unable to reach an applicable WUP Region Volume through other WUP Practices (in which case the Parties will Meet and Confer), or an Eligible Landowner agrees to retire additional acres as follows.
 - **3.18.1.** Upper Sprague WUP Region: 4,140 acres.
 - 3.18.2. Sycan WUP Region: 630 acres.
 - 3.18.3. Lower Sprague WUP Region: 5,652 acres.
 - 3.18.4. Middle Williamson WUP Region: 198 acres.
 - **3.18.5.** Lower Williamson WUP Region (including Modoc Point Irrigation District): 1,620 acres.
 - 3.18.6. Wood Valley WUP Region: 5,760 acres.
- 3.19. Practicably Irrigable Acres and Other Water Rights that Are Not Fully Developed. Practicably Irrigable Acres (PIA) rights and water right permits that are not fully developed within the SIF Regions are eligible for the WUP. Such water right retirements will not count toward the WUP Region Volumes unless the lands were already being irrigated as of December 31, 2001. If there is Water Use Development (including PIA rights), there will be a debit in the WUP Ledger for the amount of increased Net Consumptive Use. Any debit to the WUP Ledger will be based on the WUP Guidelines.
- **3.20. Dewatering.** The Parties agree that they will not take actions to dewater perennial streams within the Off-Project Area. Dewatering is defined as a decrease in stream flow resulting from irrigation withdrawals within a perennial stream reach that results in a stream flow below twenty percent of the flow at the upstream end of that perennial stream reach. In addition to the factors in subsection 3.5.7, the WUP will prioritize funding of WUP Practices to avoid dewatering, including but not limited to use of the Oregon Conserved Water Statute. Based upon the recommendation of the LE and JME Technical Team, the JME will establish a methodology to provide the measurements needed for this provision.

- **3.21. Livestock Watering.** The Parties agree to cooperate in establishing alternate, off-channel, facilities for watering livestock, including but not limited to wells. In addition, the Parties agree to cooperate in working to make existing livestock watering facilities more efficient. Retired groundwater sources may be used for exempt stockwatering use and will retain the priority date that the well was constructed.
- 3.22. Preliminary Direction to the JME and LE. The LE will obtain statements of intent from Eligible Landowners who are interested in participating in the WUP. The statements of intent will include landowner identification information, acreages, water rights, locations, SIF Region, and proposed WUP Practice(s). The LE will evaluate statements of intent prior to acceptance into the WUP to ensure compliance with the WUP requirements and priorities, and to estimate the amount of water to be realized instream resulting from reduced Net Consumptive Use. Permanent transfers of a surface water right to instream use that are funded through the WUP Program should occur only after such transfers have been prioritized as described in subsection 3.5.7, and the LE has conferred with the Klamath Tribes, the Funding Entities and the JME to verify that the priorities were followed.
- **3.23. Klamath Reclamation Project.** Non-Federal Parties will support good faith discussions to resolve potential conflicts between the Klamath Reclamation Project and the Non-Federal Parties over the management and use of water in the Upper Klamath Basin. The Parties have worked to maintain the benefits of the KBRA to the Klamath Reclamation Project.
- **3.24. Alteration of Law.** The Parties agree that if Oregon law applicable to the regulation of surface or groundwater rights in the Off-Project Area is materially altered from that in existence on December 2, 2013, except for those changes expressly contemplated and provided for in this Agreement, a Party who believes that such change in the law materially and adversely affects its position may initiate the Meet and Confer procedures of section 11 to attempt to resolve the matter.
- **3.25. Alteration of Procedures.** The Parties agree that if the procedures for enforcing a call for regulation are altered from those set out in the General Guidance to Address A "Call for Regulation" attached as Exhibit G to this Agreement, in a manner that impairs the bargained-for benefits of a Party, that Party may initiate Meet and Confer procedures to attempt to resolve the matter.

4. RIPARIAN PROGRAM

- 4.1. Overall Administration of the Riparian Program. The JME will oversee the design and overall implementation of the Riparian Program described in this section. The JME will solicit input from Funding Entities in performing these duties. The LE will be responsible for negotiating and administering Riparian Management Agreements that carry out the Riparian Program. The LE will oversee the performance and compliance requirements of the Riparian Agreements set forth in subsections 4.2 to 4.12 of this section. The respective roles of the JME, the LE and the Klamath Tribes are described in more detail in sections 7 and 8 of this Agreement.
- **4.2. Riparian Program Outcomes.** The Riparian Program will achieve the following outcomes in accordance with the Riparian Guidelines, as described in Exhibit H.
 - **4.2.1.** The overarching, long-term, outcome of the Riparian Program is to re-establish and/or maintain the full expression of successional dynamics of the riparian plant community within Riparian Management Corridors, thereby improving and maintaining water quality and fish habitat.
 - **4.2.2.** The outcome described in subsection 4.2.1 will be achieved in part by the LE and the Eligible Riparian Landowners attaining and maintaining Proper Functioning Conditions (PFC) as specified in each Riparian Management Agreement. The Parties recognize that it will take time to achieve PFC in many locations.
 - **4.2.3.** If a particular Riparian Management Area is non-functional or functional-at risk, as described in Exhibit H, the near-term desired outcome is to achieve an upward trend toward a sustainable PFC.
 - **4.2.4.** The LE will enroll Eligible Riparian Landowners in Riparian Management Agreements covering at least 80 percent of the length of the Riparian Management Corridors owned by Eligible Riparian Landowners within each SIF Region (see subsection 4.9., however, describing the circumstances under which this length criterion may be reduced to as low as 77 percent). The LE will be the lead entity in developing site specific Riparian Management Agreements and in enrolling Eligible Riparian Landowners.
 - **4.2.5.** The Westside Restoration Area is a unique region in this Agreement. For purposes of the WUP, the Westside Restoration Area is part of the Wood Valley WUP Region, and will contribute to the Wood

Valley WUP Region Volume. However, the Westside Restoration Area is not within a SIF Region, and, therefore, is not subject to calls to a SIF-based Call Threshold. The Westside Restoration Area is eligible to participate in the Riparian Program, and the regulatory assurances found in section 9 are applicable to this area.

4.2.6. The Riparian Program will also be designed to maintain viable agricultural activities within the Off-Project Restoration Area and Westside Restoration Area and with consideration of the ranching and farming infrastructure to the extent described in the Riparian Guidelines.

4.3. Categories of Areas in the Riparian Program.

- **4.3.1.** Riparian Management Areas. The Riparian Management Area is the primary focus of the management measures and restoration actions contained in a Riparian Management Agreement.
- **4.3.2.** Adjacent Transition Areas. Adjacent Transition Areas are the portions of a field adjacent to a Riparian Management Area and in the same parcel, where vegetation is managed to benefit the function of the Riparian Management Area, as specified in the Riparian Management Agreement and as described in the Riparian Guidelines.
- **4.3.3.** Non-irrigated Riparian Areas. Non-irrigated Riparian Areas are lands along a perennial stream in the Off-Project Area that are adjacent to a Riparian Management Area, that are within the same parcel, and that are currently zoned for Agricultural Use as shown in Exhibit I, but that are not covered by an irrigation water right. In these areas, livestock will be managed to limit adverse impacts to the riparian area of the stream through the use of off-site livestock watering, salt and mineral placement, and other management tools as specified in the Riparian Management Agreement. Non-irrigated Riparian Areas are not counted for purposes of determining Sufficient Participation.

4.4. Monitoring and Adaptive Management in the Three Categories of Areas.

4.4.1. Within Riparian Management Areas, the trend and condition of riparian plant communities will be initially assessed by the Proper Functioning Condition Monitoring Team, described in subsection 4.5.6, using the Proper Functioning Condition (PFC) methodology, and then subsequently monitored using Green Line, Multiple Indicator Monitoring, or another similar method that is compatible with the PFC methodology. PFC assessments will evaluate hydrologic, vegetative, and erosional/depositional conditions in the area, and produce ratings

that express the degree of resilience to relatively high flow events. In general, the ratings will indicate whether the area is in a PFC, Functional-at-Risk with an upward or downward trend, or Non-Functional category. The PFC assessment will identify factors affecting conditions that are outside of the control of the Eligible Riparian Landowner. Subsequent to the initial PFC assessment, monitoring methods compatible with the PFC methodology will track riparian conditions. If monitoring indicates that existing management measures and restoration actions are not achieving PFC or an upward trend, then the responsible landowner, the LEs and the JME Technical Team will identify what management changes, if any, are practicable and necessary to move the Riparian Management Area into these conditions.

- **4.4.2.** Within Adjacent Transition Areas, conditions will be monitored to determine whether they are meeting the applicable criteria in the Riparian Management Agreement for vegetation density and height, wetlands, and stock watering. If the monitoring shows that management is not achieving the applicable criteria, then the landowner, the LE and the JME Technical Team will collaboratively identify what changes, if any, in management are practicable and necessary for the conditions to achieve the applicable criteria.
- **4.4.3.** Within Non-irrigated Riparian Areas, conditions will be monitored to determine whether adverse impacts to the Riparian Management Corridor are being limited, as specified in the applicable Riparian Management Agreement. If the monitoring shows that management is not achieving the applicable criteria, then the landowner, the LE, and the JME Technical Team will identify what changes in management, if any, are practicable and necessary for the conditions to achieve the applicable criteria.
- **4.4.4.** Significant management changes identified in subsections 4.4.1 to 4.4.3 that are not covered by the existing Riparian Management Agreement will be incorporated into the applicable Riparian Management Agreement upon recommendation by the JME Technical Team and approval by the JME as a Major Decision, as well as by the landowner. Monitoring provisions to evaluate outcomes of management actions are described in the Riparian Guidelines.
- **4.5. Riparian Management Agreements and Riparian Action Plan.** The Riparian Management Program will be implemented through Riparian Management Agreements entered into on a willing basis between Eligible Riparian Landowners and the LE to achieve the outcomes described in subsections 4.2 through 4.4. If an Eligible

Riparian Landowner is interested in participating in the Riparian Program, the LE and the Eligible Riparian Landowner will develop a draft Riparian Management Agreement based on the Riparian Guidelines, a template form of Riparian Management Agreement approved by the JME, and, the specific characteristics, management requirements, and ecosystem restoration needs of the Riparian Management Area and the associated waterway. The Riparian Management Agreements will contain performance goals and objectives as described in subsections 4.2 through 4.4 of this section and the Riparian Guidelines. In addition, the agreements will include the following elements in which all references to width apply to one side of a stream.

- **4.5.1.** Delineation of the specific Riparian Management Area and Adjacent Transition Area, and any other area where riparian management measures and restoration activities (if any) are to occur. The Riparian Management Area and Adjacent Transition Area will be consistent with the Riparian Guidelines.
- **4.5.1.1.** The minimum allowable width of the Riparian Management Areas will be the lesser of 50 feet or a reasonably consistent contour 2 feet above the elevation of the water surface, in the 2005 LiDAR digital elevation model (or other mutually agreed upon elevation baseline), constrained by an absolute minimum of 30 feet, as specified in more detail in the Riparian Guidelines.
- **4.5.1.2.** The maximum required width of the Riparian Management Areas is either 100 feet or 130 feet, depending on which of the options described in subsection 4.5.2 the landowner elects.
- **4.5.2.** Landowners may elect either of two options for Riparian Management Agreements in terms of the width of the Riparian Management Area and the Adjacent Transition Area.
- **4.5.2.1.** Under Option 1, the minimum width of the Riparian Management Area is the lesser of 50 feet or a reasonably consistent contour two feet above the elevation of the water surface in the 2005 LiDAR digital elevation model (or other mutually agreed upon elevation baseline). The absolute minimum width of the Riparian Management Area under Option 1, regardless of slope, is 30 feet, unless one of the exceptions set forth in the Riparian Guidelines applies. The maximum required width of a Riparian Management Area under Option 1 is 100 feet. Within these limits, a baseline width of 75 feet will be used as a starting point for delineation, and will be modified to reflect site conditions according to the Riparian Guidelines. The Adjacent

Transition Area will encompass the area in fields adjacent to the Riparian Management Area such that the combined width of the two areas is 130 feet, unless the landowner agrees to a greater width.

- **4.5.2.2.** Under Option 2, there is no requirement for an Adjacent Transition Area. Instead, there is a wider Riparian Management Area. A 90 foot baseline width for the Riparian Management Area will be used as a starting point for delineation, and will be modified to reflect site conditions according to the Riparian Guideline. The maximum required width will be 130 feet. Otherwise, the widths will be decided based on the same criteria as for Option 1.
- **4.5.3.** The Riparian Management Agreements will prescribe management measures associated with the Riparian Management Areas and, if applicable, the Adjacent Transition Areas. The measures will be designed to achieve the outcome described in subsection 4.2.1 by establishing and maintaining PFC through various land management methods which may include grazing and vegetation management as well as ecosystem restoration actions.
- **4.5.4.** If the Eligible Riparian Landowner owns non-irrigated lands along a perennial stream that are zoned for Agricultural Use, and that are within the same parcel as the Riparian Management Area, the Riparian Management Agreement will identify a non-irrigated Riparian Management Area. These lands are not included for purposes of determining Sufficient Participation. However, the Riparian Management Agreement will include management measures to limit adverse impacts to the riparian area through the use of off-site livestock watering, salt/mineral placement, and other management tools.
- **4.5.5.** The Riparian Management Agreement will identify the person(s) responsible for carrying out each of the specified management measures and/or restoration actions, including who is responsible for paying any up-front/capital costs needed to implement the measures and/or actions. If actions or measures require approval of Funding Entities, the Eligible Riparian Landowner's obligations under the agreement will be contingent upon receipt of such funding. In general, Eligible Riparian Landowners will be responsible for ongoing management measures and maintenance of improvements made under a Riparian Management Agreement (excluding catastrophic loss), but will not be responsible for the initial costs of any restoration actions (including new fencing). To the extent that management measures will demonstrably and permanently reduce the profitability of a landowner's

operations, the agreement may provide for the landowner to be compensated accordingly.

- **4.5.6.** Each Riparian Management Agreement will identify monitoring and reporting requirements, consistent with the Riparian Guidelines, including provisions providing access for monitoring purposes. Those requirements will include required reporting to the JME by the LE, as well as reporting requirements for the Eligible Riparian Landowner. Monitoring will be carried out by the PFC Monitoring Team, made up of appropriate technical expertise. The PFC Monitoring Team will include at least one technical representative of the LE, one technical representative of the Klamath Tribes, and one technical representative each of a Federal and State agency if they so desire. Different substantive disciplines will be represented on the PFC Monitoring Team, depending on the nature of the lands and restoration or management actions being monitored. The frequency of monitoring will be determined based on the guidelines contained in Exhibit H. All data, analyses, and reports generated by these monitoring efforts will be made available to the JME in a Timely manner. Substantive changes in the approach to monitoring described in Exhibit H can be made as a Major Decision by the JME.
- **4.5.7.** Each Riparian Management Agreement will identify how the obligations in the Riparian Management Agreement can be enforced. In general, the agreement will provide that in the event of a failure to perform the Eligible Riparian Landowner's obligations, the LE will first give notice to the landowner of the default. If the Eligible Riparian Landowner fails to cure the default within the time specified in the agreement, the LE must then give notice to the JME and the Klamath Tribes, and thereafter the JME and/or the Klamath Tribes may perform the obligation directly and obtain reimbursement for its expenses from the Eligible Riparian Landowner. If the default continues for a period of more than 60 days after notice to the landowner, the agreement will provide that the Klamath Tribes and the United States each have the right to enforce the agreement.
- **4.5.8.** The JME will establish and maintain an Upper Klamath Basin Riparian Management and Restoration Action Plan (the Riparian Action Plan) that identifies restoration actions, geographic priorities and funding sources for the Riparian Program at the scale of the SIF Region. Establishment and amendment of the Riparian Action Plan is a Major Decision of the JME. The Riparian Action Plan will include restoration actions that should be included in specific Riparian Management Agreements.

- **4.5.9.** When a Riparian Management Agreement is developed that contains a restoration action, conceptual design of the restoration action will be scoped with the landowner, LE, and JME Technical Team. Design and planning of restoration actions at the specific project scale may be incorporated into Riparian Management Agreements.
- **4.5.10.** Design and planning of restoration actions will be done by a project team in close coordination with the JME Technical Team, for review and approval by the JME. The project team may include the Klamath Tribes, the United States, the LE, the State, and other appropriate interests. Such actions will require interactions with a diverse array of scientific and engineering specialists, as well as consultation with non-parties.
- **4.5.11.** Restoration actions identified in Riparian Management Agreements may include, but are not limited to:
- **4.5.11.1.** Treatment or elimination of surface return flows, which may involve activities outside of a Riparian Management Area;
 - **4.5.11.2.** Removal or breaching of levees and dikes;
 - 4.5.11.3. Stream channel realignment; and
- **4.5.11.4.** Installation of fish screens, which may involve activities outside of a Riparian Management Area.
- **4.5.12.** Design and planning of generally applicable management measures that may be incorporated into Riparian Management Agreements will be done collaboratively by the Klamath Tribes, the United States, and the LE. Results of the design and planning effort will be used by the LE for negotiating draft Riparian Management Agreements with Eligible Riparian Landowners, and by the JME Technical Team and the JME when reviewing the agreements and considering them for approval. Management measures identified in Riparian Management Agreements may include, but are not limited to:
- **4.5.12.1.** Vegetation management, including associated grazing management;
- **4.5.12.2.** New and existing fencing; existing riparian fencing that does not meet the Riparian Guidelines but that is not contributing to a downward trend in PFC may be incorporated (on a case by case basis) as a part of the management strategy and remain in place for its useful life;

- 4.5.12.3. Alternate livestock watering facilities; and
- **4.5.12.4.** Irrigation practices.
- 4.6. Approval of Agreements. Each draft Riparian Management Agreement must be reviewed by the JME Technical Team, which will evaluate the technical adequacy of the proposed management and restoration measures, as well as the extent to which the Riparian Guidelines were followed. The JME Technical Team will make recommendations to the JME on these issues. Then the JME will confirm whether the supporting information is adequate, and whether the Riparian Guidelines were followed. Following this procedure, each Riparian Management Agreement must be approved by the JME, including the specific approval of the Klamath Tribes and the United States, as well as by the LE and the Eligible Riparian Landowner, before it becomes final. The Parties acknowledge that collaborative relationships among the Klamath Tribes, the United States, the LE, and Eligible Riparian Landowners will increase the probability of the JME rapidly approving Riparian Management Agreements.
- **4.7. Amendment of Riparian Guidelines.** The Riparian Guidelines contained in Exhibit H to this Agreement may be amended by the JME as a Major Decision.
- 4.8. Sufficient Participation. Within each SIF Region, the minimum threshold for Eligible Riparian Landowner participation in the Riparian Program is 80% of the length of the Riparian Management Corridor owned by Eligible Riparian Landowners, subject to possible deviations specified in subsection 4.9. "Sufficient Participation" means that this minimum length of the Riparian Management Corridor is covered by Riparian Management Agreements and that the Eligible Riparian Landowners are in material compliance with the terms of such Riparian Management Agreements. Material non-compliance equates to non-participation under subsection 4.12. The length required for Sufficient Participation in each SIF Region shall be determined by the JME Technical Team based on conditions existing as of the Effective Date of this Agreement.
- **4.9. Deviation from Sufficient Participation Criterion.** The Parties acknowledge that in some cases the layout of parcels along the Riparian Management Corridor owned by Eligible Riparian Landowners is such that a Sufficient Participation level of 80% may not be exactly attainable. For example, a participation level of 79% might jump to 82% if the smallest remaining non-participating landowner was to

enroll. Therefore, if participation in Riparian Management Agreements is at least 77%, then the LE may petition the JME to accept a level of Sufficient Participation less than 80% (but not less than 77%). Approval of such a petition would be a Major Decision of the JME.

- 4.10. Permanence of Riparian Management Agreements and Transfer of Ownership. Notwithstanding the Transition Period provisions in subsection 5.4, the Riparian Management Agreements are to be permanent. To this effect, each Riparian Management Agreement, or a memorandum summarizing the agreement, must be recorded in the deed records of Klamath County within 30 days of the date the agreement is approved by the JME. The agreement must require that the landowner provide Notice to the LE when the Riparian Management Area, the Adjacent Transition Area, or the Non-Irrigated Riparian Area is transferred to a new owner. When a transfer occurs, the LE will contact the new owner to provide information regarding the agreement.
- **4.11. Relation between the Riparian Program and the Water Use Program.** Failure to achieve and maintain Sufficient Participation within a SIF Region may result in an increase in the Call Threshold for that SIF Region, as described in subsection 3.10 of this Agreement.
- **4.12. RMA Ledger.** Within each SIF Region, the linear extent of the Riparian Management Corridor enrolled in and compliant with Riparian Management Agreements will be tracked in the RMA Ledger. The RMA Ledger will be managed in the same manner specified in section 3 for the WUP Ledger. The LE will make draft entries into the RMA Ledger and propose amendments to approved entries, and will develop the supporting information. Draft entries and proposed amendments to approved entries will be reviewed by the JME, and approved as a Major Decision or denied after the JME considers advice from the JME Technical Team. At the beginning of each month, approved entries in the RMA Ledger will be the basis for the calculation of the Type B SIF Adjustment specified in subsection 3.10, as part of the calculation of the Call Threshold for that month. Approved entries in the RMA Ledger will also be the basis for determinations regarding attainment of Sufficient Participation in subsections 4.8 and 4.9, and of the related participation and regulatory thresholds during the Transition Period specified in subsections 5.4 and 5.5.

5. TRANSITION PERIOD

Introduction. This Agreement is dependent upon acceptance by a wide number of Eligible Landowners and Eligible Riparian Landowners in the Off-Project Area and Off-Project Restoration Area, as well as by the Klamath Tribes and other governmental bodies, and contains a number of contingencies, which are set forth in section 10. The Parties recognize that it will take time to develop and implement the Water Use and Riparian Programs described in sections 3 and 4 of this Agreement. There will be a Transition Period of five-years, ending on March 31, 2019, during which the transitional provisions described in this section will apply. During this Transition Period, the Parties intend that Water Use Agreements securing the WUP Region Volumes described in section 3, and Riparian Management Agreements, securing the management measures and restoration actions described in section 4, will be entered into. However, the Parties also acknowledge that the contingencies set forth in section 10, including the funding to make the Water Use Agreements and the Riparian Management Agreements permanent. likely will occur over a longer period of time. As a result, there are effectively two transition periods, the first one of five years during which the Water Use Agreements securing the WUP Region Volumes are entered into, and the Riparian Management Agreements securing the management measures are entered into, and then a second period during which the long-term funding required to make these agreements permanent and to meet the other contingencies described in section 10 is provided.

In addition, the Parties also acknowledge that a critical element of success for both the WUP and the Riparian Program is the funding and development of the capacity in the JME, the LEs, and the Klamath Tribes needed for implementation and administration of the programs.

- 5.2. Initial Administration. Following the Effective Date, the Non-Federal and non-State Parties will work to form the JME and the LE as soon as possible, and preferably within 30 days. Once the JME and LE are formed, they will work with the USFWS, the Oregon Watershed Enhancement Board (OWEB), the Natural Resources Conservation Service (NRCS), the Deschutes River Conservancy (DRC), and the National Fish and Wildlife Foundation (NFWF) to begin implementation of the transitional WUP and the transitional Riparian Program. Without limiting the generality of the foregoing, the JME and LE will:
 - **5.2.1.** Develop one or more templates for Water Use Agreements and for Riparian Management Agreements;

- **5.2.2.** Develop funding sources for the WUP and for the Riparian Program;
- **5.2.3.** Work with the NFWF, DRC, OWEB, NRCS, USFWS, and the JME Technical Team to begin enrolling Eligible Landowners in the WUP and Eligible Riparian Landowners in the Riparian Program as described in subsections 5.3 and 5.4.

5.3. Transitional Water Use Program

- **5.3.1. Year 1.** By June 1, 2014, the LE, working with the JME, OWEB, NRCS, NFWF, DRC and USFWS as applicable, will enter into Water Use Agreements with Eligible Landowners to reduce Net Consumptive Use. The total amount of reduction of Net Consumptive Use from Water Use Agreements and other WUP Practices will be at least 5,000 acre-feet in the Off-Project Area relative to Baseline Conditions. The Water Use Agreements may include contingencies consistent with the contingencies in this Agreement, but must be for a period of at least three to five years, and must include the option for permanent or longer term reductions of Net Consumptive Use. A Water Use Agreement generally should not be used to pay an Eligible Landowner for the reduction of Net Consumptive Use where the applicable water right is already being regulated in that year, except where the Water Use Agreement provides for long-term or permanent reductions in water use. The LE and the JME will develop the WUP Ledger and begin evaluating WUP Ledger entries as early as possible in order to assure that the JME is able to make a decision by June 1, 2014, regarding whether the 5,000 acre-foot WUP Volume has been met. For purposes of regulation under subsections 5.5.1 during Year 1, compliance with executed Water Use Agreements and other WUP Practices must be maintained at levels sufficient to produce the WUP Volume of 5,000 acre-feet.
- **5.3.2. Year 2.** By March 31, 2015, the LE will enter into statements of intent for Water Use Agreements with Eligible Landowners or other WUP Practices to reduce Net Consumptive Use by an additional 6,250 acrefeet beyond the amount described in subsection 5.3.1. The statements of intent must be converted to Water Use Agreements or other WUP Practices and approved by the JME by March 31st, 2016. For purposes of regulation under subsection 5.5.2 during Year 2, compliance with executed Water Use Agreements or other WUP Practices must be maintained at levels sufficient to produce a WUP Volume of at least 5,000 acre-feet.
- **5.3.3. Years 3-5.** By March 31st of each year between 2016 and 2018, the LE will enter into statements of intent for Water Use Agreements with

Eligible Landowners or other WUP Practices to reduce Net Consumptive Use by an additional 6,250 acre-feet each year beyond the amounts described in subsections 5.3.1 and 5.3.2, proportionally allocated among the WUP Region Volumes. The statements of intent must be converted to Water Use Agreements or other WUP Practices and approved by the JME by March 31st of the following year. Each WUP Region Volume, and the Total WUP Volume of 30,000 acre-feet, must be attained (consistent with section 3) by March 31, 2019. For purposes of regulation under subsection 5.5.3, compliance with executed Water Use Agreements or other WUP Practices must be maintained at levels sufficient to produce a WUP Volume of at least 11,250 acre-feet during the 2016 Irrigation Season, 17,500 acre-feet during the 2017 Irrigation Season, 23,750 acre-feet during the 2018 Irrigation Season, and 30,000 for all subsequent Irrigation Seasons.

5.3.4. The proportional allocation among WUP Region Volumes in subsection 5.3.3 is illustrated by the following example. If a specific WUP Region had a WUP Region Volume obligation of 6,000 acre-feet, its annual obligations during years 3-5 of the Transition Period would be calculated as follows. During years 1-2, a total volume of 11,250 acrefeet must have been produced by all of the WUP Regions combined. Each year thereafter, the obligation increases by 6,250 acre-feet, so that in year 3 the total obligation for all of the WUP Regions is 17,500 acrefeet, which is 58.33% of 30,000 acre-feet. By March 31^{st} of year 3 the LE would enter into statements of intent for 3,500 acre-feet for that WUP Region $(6,000 \times 58.33\% = 3,500 \times 58.33\% =$

5.4. Transitional Riparian Program.

5.4.1. Year 1. By June 1, 2014, the LE will enter into statements of intent for Riparian Management Agreements with Eligible Riparian Landowners for at least fifty percent of the total length of the Riparian Management Corridor summed across all SIF Regions within the Off-Project Restoration Area. By March 31, 2015, at least ten Riparian Management Agreements will be executed. These may be short-term transitional agreements, or permanent agreements. Riparian Management Agreement will provide that planning and implementation of management measures will be started as soon as the Riparian Management Agreement is signed. Specific restoration actions may not be sufficiently developed to be included in a permanent Riparian Management Agreement until the Riparian Action Plan is completed for a particular area. In such cases, short-term transitional Riparian

Management Agreements may be utilized, but they must contain an option for a permanent agreement, which may be subject to contingencies consistent with this Agreement. During the Transition Period, transitional Riparian Management Agreements will be credited towards the participation levels specified in subsection 5.4. For purposes of regulation under subsection 5.5.2, compliance with at least ten executed Riparian Management Agreements must be maintained after March 31, 2015.

- **5.4.2. Year 2.** By March 31, 2015, the LE will enter into statements of intent for Riparian Management Agreements with Eligible Riparian Landowners encompassing at least 58% of the total length of the Riparian Management Corridor summed across all SIF Regions within the Off-Project Restoration Area. Prior to March 31, 2016, executed Riparian Management Agreements that have been approved by the JME must encompass at least 40% of the total length of the Riparian Management Corridor summed across all SIF Regions within the Off-Project Restoration Area. For purposes of regulation under subsection 5.5.3, compliance with executed Riparian Management Agreements must be maintained at levels encompassing at least 40% of the total length of the Riparian Management Corridor summed across all SIF Regions within the Off-Project Restoration Area after March 31, 2016.
- **5.4.3. Years 3-5.** By March 31st of each year between 2016 and 2018, the LE will enter into statements of intent for Riparian Management Agreements with Eligible Riparian Landowners encompassing at least 66% (in 2016), 73% (in 2017), or 80% (in 2018) of the Riparian Management Corridors within each SIF Region. Prior to March 31st of each year between 2017 and 2019, the LE will enter into Riparian Management Agreements with Eligible Riparian Landowners that have been approved by the JME, and encompass at least 60% (in 2017), 70% (in 2018), or 80% (in 2019, and subject to subsection 4.8) of the Riparian Management Corridors within each SIF Region. For purposes of regulation under subsection 5.5, compliance with executed Riparian Management Agreements must be maintained at levels encompassing at least 60% (after March 31, 2017), 70% (after March 31, 2018), or 80% (after March 31, 2019, and subject to subsection 4.8) of the Riparian Management Corridors within each SIF Region.
- **5.4.4**. Riparian Management Agreements may be entered into along Riparian Management Corridors within the Westside Restoration Area.
- 5.5. Regulation of Water Use During the Transition Period.

- **5.5.1.** During the Irrigation Season in 2014, prior to June 1st, the Klamath Tribes and the BIA may call to the Long-Term SIF Thresholds. After June 1st, the Klamath Tribes and the BIA may call to the Long-Term SIF Thresholds if the conditions set forth in subsections 5.3.1 and 5.4.1 have been met by June 1st. If the 2014 conditions are not met by June 1st and maintained thereafter, then the Klamath Tribes and the BIA may call to the Tribal Water Rights until the conditions are met.
- **5.5.2.** During the Irrigation Season in 2015, the Klamath Tribes and the BIA may call to the Long-Term SIF Thresholds if compliance with Water Use Agreements has been maintained as specified in the last sentence of subsection 5.3.2, if the Year 2 conditions set forth in subsection 5.3.2 have been met, and if compliance is maintained with the terms of the Riparian Management Agreements as specified by the last sentence in subsection 5.4.1. If these conditions are not met, then the Klamath Tribes and the BIA may call to the Tribal Water Rights until the conditions are met.
- **5.5.3.** During the Irrigation Seasons from 2016 to 2018, within each WUP Region, the Klamath Tribes and the BIA may call to the applicable Long-Term SIF Threshold(s) if that WUP Region has met the conditions specified in subsections 5.3.3 and 5.3.4. Within each SIF Region, the Klamath Tribes and the BIA may call to the applicable Long-Term SIF Threshold if compliance is maintained with the terms of the Riparian Management Agreements as specified by the last sentence in subsection 5.4.2. If these conditions are not met, then the Klamath Tribes and the BIA may call to the Tribal Water Rights until the conditions are met within the applicable WUP Region or SIF Region.
- **5.5.4.** Commencing in Year 3, the Klamath Tribes and the BIA may call to the Tribal Water Right for lake levels in Upper Klamath Lake during the Transition Period if a WUP Region Volume (adjusted to reflect the phasing of the volumes set forth in subsection 5.3.3.), is not met, in which case regulation may proceed according to subsection 3.8.2 and 3.8.3.

6. ADDITIONAL AGREEMENTS

- 6.1. Access for Exercise of Treaty Rights. The State agrees to seek funding from the Oregon Parks and Recreation Department, the USFWS, and other sources for public access to at least four river sites, one on the Wood River, one on the Williamson River, one on the Sycan River, and one on the Sprague River that are suitable for exercise of tribal fishing rights and other public purposes. The State agrees to work collaboratively with the Klamath Tribes to (1) locate each site, (2) agree on how each site will be managed and maintained, and (3) identify funding within five years of the Effective Date. The State also will work with the LE to explore acquisition of such sites on a willing seller basis.
- **6.2. KBRA and KHSA.** The Parties, other than the United States, agree mutually to Timely promote, support, strive, and use Best Efforts to obtain funding and authorizations necessary to implement the KBRA and this Agreement. The Parties, other than the United States, also agree that they will not oppose authorization and implementation of the KBRA or the KHSA including any legislation required to authorize and implement those agreements.
- **6.3. Klamath Allottees.** Nothing in this Agreement shall be construed to limit or remove existing water rights of Klamath Allottees.
- **6.4. Walton Rights.** Nothing in this Agreement shall be construed to limit or remove existing *Walton* rights.

7. JOINT MANAGEMENT ENTITY

- 7.1. Formation. The Parties agree to form a JME as an Oregon tax-exempt non-profit corporation. The JME will include a Board of Directors (Board) responsible for decision-making, and shall appoint a JME Technical Team to conduct analyses, gather information, and make recommendations to the Board. The JME may employ one or more staff to assist in carrying out its functions.
 - **7.1.1.** The Board will include one voting director representing each of the following: the Klamath Tribes, the United States, the State, and the LE.
 - **7.1.2.** The Klamath Tribes will appoint their voting director and two other non-voting directors. The three Klamath Tribes directors will collectively decide how their voting director votes.
 - **7.1.3.** The LE representation on the JME will be comprised of a voting director, and six non-voting directors. There will be one director from each of the six WUP Regions, and one representing the Allottees. At least one of the directors shall be a member of the Upper Klamath Water Users Association. The seven LE directors will collectively decide how the voting director votes.
 - **7.1.4.** Until legislation is enacted authorizing the State to participate as a voting member of the JME, the State will appoint three non-voting representatives to make recommendations to the JME. When legislation is enacted authorizing State agencies to participate as a voting member of the JME, the State will appoint its voting director, and two other non-voting directors. The three State directors will collectively decide how the voting director votes.
 - **7.1.5.** Until Federal legislation is enacted authorizing the United States to participate as a voting member of the JME, the United States will appoint five non-voting representatives (as consistent with existing authorities), including a designee of the Secretary and representatives of the BIA, National Marine Fisheries Service (NMFS), USFWS, and USGS. When Federal legislation is enacted authorizing the United States to participate as a voting member in the JME, the United States will appoint the Secretary's designee as the voting director and four non-voting directors, representing the BIA, NMFS, USFWS, and the USGS, and the five Federal directors will collectively decide how the voting director votes.

- **7.1.6.** The JME will assume the obligations of the Upper Basin Team for purposes of the KBRA. The USFWS is the "Federal Lead Party" for purposes of Section 16 of the KBRA, and as such must provide oversight for the expenditure of Federal funding for the WUP, to the extent that the funding is provided under the KBRA.
- **7.2. Functions.** The JME will have overall responsibility for implementation of this Agreement including the design, development and oversight of the WUP described in section 3, the Riparian Program described in section 4, and the Transitional Water Use and Transitional Riparian Programs described in section 5. Without limiting the generality of the foregoing, the JME will carry out the following functions:
 - **7.2.1.** Receive funding, and administer and coordinate funding-related activities necessary for the JME to fulfill its role in implementing this Agreement;
 - **7.2.2.** Prioritize expenditures of JME funds to implement the WUP, and the Riparian Program, which may occur through development of collaborative plans for implementation of those programs;
 - **7.2.3.** Develop template forms of Water Use Agreements and of Riparian Management Agreements;
 - **7.2.4.** Review and approve Water Use Agreements and Riparian Management Agreements;
 - **7.2.5.** Review and approve WUP Ledger entries, use the WUP Ledger to track the WUP Region Volumes over time, and determine whether the WUP Region Volumes have been attained at the end of the Transition Period and maintained thereafter, as well as whether the WUP Region Volume thresholds established for the Transition Period have been attained;
 - **7.2.6.** Assure that WUP Practices are consistent with subsections 3.12 and 3.13 and accounted for under subsection 3.7;
 - **7.2.7.** Maintain a JME Technical Team;
 - **7.2.8.** Develop, maintain, and amend the WUP Guidelines as may be necessary and as provided in this Agreement;
 - **7.2.9.** Perform the functions in the KBRA specified for the Upper Basin Team;

- **7.2.10.** Review and approve entries in the RMA Ledger, use the RMA Ledger to track the extent to which Eligible Riparian Landowners are participating in and are in compliance with Riparian Management Agreements within SIF Regions over time, and determine whether Sufficient Participation has been attained at the end of the Transition Period and maintained thereafter, as well as whether the related thresholds established for the Transition Period have been attained; and
- **7.2.11.** Enter into and enforce contractual agreements as necessary to fulfill its role in the implementation of this Agreement.

7.3. Decision-making.

- **7.3.1.** Decisions to approve, disapprove, or amend Water Use Agreements, the Riparian Management Agreements, the WUP Guidelines, the Riparian Guidelines, the Riparian Action Plan, entries into the WUP and RMA ledgers, appointments to the JME Technical Team, and proposed deviations from Sufficient Participation are "Major Decisions." Major Decisions require the unanimous affirmative vote of the voting directors of the Klamath Tribes, the United States, and the LE. If there is not an affirmative vote of these parties, then the JME will follow the dispute resolution procedure set forth in subsection 7.4 of this section.
- **7.3.2.** The development and amendment of the Articles of Incorporation and Bylaws of the JME will be by unanimous written agreement of the Klamath Tribes, the United States, the State, and the LE.
- **7.3.3.** Each voting director's vote on the JME will be determined by a separate vote of the entity's voting director and non-voting directors based on the interests of the entity they represent and consistent with the terms of this Agreement. If a non-voting director is absent, the proxy for that vote is assigned to the voting director.
- **7.3.4.** Other than Major Decisions, the JME will make decisions by Consensus of the voting directors.

7.4. Dispute Resolution.

7.4.1. Each voting director shall make Best Efforts to reach Consensus on every matter before the Board. If a voting director opposes a proposed decision he or she must propose an alternative to resolve the matter.

- **7.4.2.** In the event that the voting directors on the Board are unable to agree on one or more matters, any non-agreeing voting director may deliver written notice to the Board, stating that a dispute exists and specifying in reasonable detail the nature of such dispute. Within thirty days following the delivery of such notice, the Board shall submit the dispute to one or more independent persons or entities for non-binding mediation. The Board may select each year one or more independent persons or entities that are acceptable to all voting directors to perform the mediation. If the Board has not pre-selected mediators at the time a notice of a dispute is delivered, or if the pre-selected mediators are not available at that time, the Board will select an independent person to perform the mediation. If the Board cannot agree on a mediator it will follow the Meet and Confer provisions in section 11.
- **7.5. Articles of Incorporation.** The Articles of Incorporation for the JME must be consistent with subsections 7.1 to 7.4, above.
- **7.6. Initial Funding.** The Non-Federal Parties expect that the JME will be formed with funding from the United States and the State, along with foundation and other sources of funding. It is the intent of the Parties that the JME have sufficient core funding to begin implementation of sections 3 through 5 of this Agreement within one year of the Effective Date.
- **7.7. Permanent Funding.** The Non-Federal Parties expect that the JME will be funded by the United States and the State, along with foundation and other sources of funding. It is the intent of the Parties that the JME has sufficient funding to fulfill its role in the implementation of this Agreement.
- **7.8. JME Technical Team.** The JME will appoint a JME Technical Team with appropriate expertise to assist in carrying out its roles and responsibilities. This will be a Major Decision of the JME. The JME Technical Team may include individuals from entities that are not Parties.
- **7.9.** Coordination Prior to Formation of the JME. Prior to formation of the JME, the USFWS, OWRD, the Klamath Tribes, and two representatives of the Parties who are irrigators in the Off-Project Area will work informally to coordinate the actions of Federal, State, tribal and private parties to implement this Agreement before the JME is formed.

8. LANDOWNER ENTITY

- 8.1. Formation. Eligible Landowners within the Off-Project Area will form the Landowner Entity (LE) to work with the JME and the Parties to implement the WUP and the Riparian Program. The LE will carry out its responsibilities concerning the WUP through an ORS Chapter 554 entity organized for the purposes of achieving the irrigation and other program benefits derived through this Agreement and to fulfill the obligations herein necessary to implement this Agreement. All holders of an irrigation or livestock water right in the Off-Project Area may be members of this entity. The Board of Directors of the LE will include one representative from each WUP Region (and may include an alternate for each), and an Allottee (and alternate). Each of these directors will serve as the LE representatives for purposes of section 7. The LE will carry out its responsibilities concerning the Riparian Program through a separate entity; however, this entity will have the same Board of Directors as the Chapter 554 entity.
- **8.2.** Powers and Duties Water Use Program. For the purposes of its role in the implementation of the WUP, the LE will have powers and duties that include, but that are not limited to:
 - **8.2.1.** As described in section 3, the LE will develop draft Water Use Agreements. Once approved by the JME, the LE will enter into, and administer Water Use Agreements for each Eligible Landowner participating in the WUP. The LE will provide outreach to Eligible Landowners, and may target such efforts to priority areas.
 - **8.2.2.** As described in section 3, the LE will obtain approval or amendment of any Water Use Agreement from the JME as a Major Decision.
 - **8.2.3.** As described in section 3, the LE will submit to the JME for review and approval any new WUP Practices or changes in WUP Practices resulting in a change in Net Consumptive Use.
 - **8.2.4.** As described in section 3, if there is a shortfall in WUP Volumes, the LE will work with Eligible Landowners to eliminate the shortfall.
 - **8.2.5.** As described in section 3, if stream flows fall below a Call Threshold, the LE will work with Eligible Landowners to correct the shortfall.
 - **8.2.6.** As described in section 3, the LE may develop a short-term water use reduction program, which may include but is not limited to water

right leasing, to address atypical conditions when Long-Term SIF Thresholds are not met.

- **8.2.7.** As described in section 3, the LE will develop the supporting information for draft entries into the WUP Ledger, and submit the draft entries for review and approval by the JME.
- **8.2.8.** The LE may work with Eligible Landowners to enter into rotation agreements or other means to achieve the objectives of this Agreement.
- **8.3. Powers and Duties Riparian Program.** For the purposes of its role in the implementation of the Riparian Program, the LE will have powers and duties that include, but that are not limited to:
 - **8.3.1.** As described in section 4, the LE will develop draft Riparian Management Agreements. Once approved by the JME, the LE will enter into, and administer Riparian Management Agreements for each Eligible Riparian Landowner participating in the Riparian Program. The LE will provide outreach to Eligible Riparian Landowners, and may target such efforts to priority areas.
 - **8.3.2.** As described in section 4, participation in monitoring and reporting riparian management and restoration activities, and enforcing Riparian Management Agreements.
 - **8.3.3.** As described in section 4, participation in monitoring and reporting Sufficient Participation in each SIF Region, and working to increase participation where needed.
 - **8.3.4.** As described in section 4, the LE will develop the supporting information for draft entries into the RMA Ledger, and submit the draft entries for review and approval by the JME.

8.4. Other Powers of the LE.

- **8.4.1.** The LE has the authority to make the determination called for under subsection 10.1.11.
- **8.4.2.** Beginning on March 31, 2015, the LE has the authority to agree to amendments of this Agreement under the process described in subsection 12.1.1.
- **8.5. Funding.** The LE may receive funding from the JME and/or from other sources. In addition, the LE formed as a water corporation under ORS chapter 554 may adopt assessments to fund its operations, pursuant to

ORS chapter 554. However, any form of assessment will require the unanimous consent of the entire Board of Directors of the LE.

9. REGULATORY ASSURANCES.

- 9.1. Non-Federal Parties who seek regulatory assurances under the KBRA and/or this Agreement (described in this Agreement as Regulatory Assurances) may qualify for coverage under an incidental take permit using General Conservation Plans (GCP) or Habitat Conservation Plans (HCP) under Section 10(a)(1)(B) of the Endangered Species Act (ESA).
- **9.2.** It is the intent of the Parties that any GCPs or HCPs will be based upon a conservation strategy for the species and address the needs of the local community.
- **9.3.** It is the intent of the Parties that Water Use Agreements and Riparian Management Agreements will contain terms and conditions that are expected to qualify for Regulatory Assurances, including coverage for individual landowners under an incidental take permit.
- 9.4. Applicants for incidental take permits may use either a GCP or HCP. The GCPs or HCPs described in this section of the Agreement will address the Non-Federal Party's actions that occur in the Off-Project Area and shall address any effects of such actions on listed species under the ESA. To the extent that GCPs and/or HCPs address effects on unlisted species, the resulting incidental take permits issued by USFWS and NMFS will clarify how the permits will cover those unlisted species if and when the species become listed under the ESA.
 - **9.4.1.** General Conservation Plan. USFWS and NMFS will lead the development of GCP(s) consistent with ESA section 10(a)(1)(B), and USFWS' and NMFS' implementing regulations and policy. The development of a GCP is undertaken by the USFWS and NMFS, rather than by an individual applicant, and is based upon the conservation strategy for the species and addresses the needs of the local community. This Agreement will provide the context for achieving these goals. Non-Federal Parties may then choose whether they wish to apply for a certification of inclusion in the incidental take permit using a GCP, or conversely, whether they prefer to develop their own plan using the HCP process. Non-Party landowners may seek inclusion in the General Conservation Plan.
 - **9.4.2.** Habitat Conservation Plan. As an alternative to the GCP (s) described above, Non-Federal Parties or the LE may develop HCP (s) consistent with the applicable elements of this Agreement for use with other information and documents necessary to apply for an incidental take permit(s) under ESA section 10(a)(1)(B) and implementing regulations for such listed species. In that case, USFWS and NMFS

shall assist in the development of any such HCP(s) and related documents by providing outreach and guidance about statutory, regulatory, and policy standards and by facilitating development of associated application packages that meet applicable standards.

- 9.5. The GCP(s) and/or HCP(s) will include all necessary components listed in USFWS and NMFS regulations and policies including, but not limited to: (1) a conservation strategy with biological goals and objectives; (2) use of adaptive management as a tool to address uncertainty in the conservation of covered species; (3) a monitoring and reporting program to provide necessary information to assess compliance, potential impacts, progress toward biological goals and objectives, and information for adaptive management; (4) opportunity for applicant and public participation; and (5) measures to address the needs of the local community. The GCP and/or HCP must otherwise meet application and issuance criteria for incidental take permits under the USFWS and NMFS implementing regulations. Many of these elements will be developed as part of the Water Use Agreements, the Riparian Management Agreements, and other pertinent aspects of this Agreement. These will be compiled and adapted for Regulatory Assurances purposes to the maximum extent practicable.
- 9.6. Activities covered under the GCP(s) and/or HCP(s) may include, but not be limited to, diversion and application of water, agricultural operations, grazing, road construction and maintenance, vegetation management, timber management, and actions associated with restoration, management, and maintenance of the riparian corridor. Measures for minimization and mitigation of incidental take under the GCP and/or HCPs will be based on USFWS' and NMFS' evaluation, in cooperation with applicants, of site-specific conditions, and may include, but not be limited to screening of diversions, management of livestock access, irrigation practices that prevent stream dewatering, protection and enhancement of riparian vegetation, fish passage improvement, culvert replacement, and reduction of erosion and sedimentation from streambanks and roads.
- 9.7. Party applicants shall support reasonable opportunities for participation of other Parties to this Agreement in development of the GCP(s) and any HCP(s). In addition to public notice and comment required by ESA section 10(c) and USFWS and NMFS implementing regulations (50 C.F.R. §§ 17.22 and 222.303, respectively) for a decision(s) on whether to issue any incidental take permit(s), USFWS and NMFS shall provide reasonable opportunities for public participation in development of the GCP and encourage Non-Federal Party applicants to include participation by interested Tribes and

provide reasonable opportunities for public participation in development of any HCP(s) consistent with USFWS and NMFS' policy. Before reaching a decision on whether to issue an incidental take permit, USFWS and NMFS shall seek and consider input from interested Tribes on each application.

- **9.8.** The GCP(s) and any HCP(s) shall include conservation and mitigation measures consistent with this Agreement and provide measures for responding to reasonably foreseeable changed circumstances to the maximum extent practicable and consistent with the ESA and NMFS and USFWS implementing regulations.
- **9.9.** The GCP(s) and/or HCP(s) developed under this Agreement is/are expected to be applicable to that area described in the KBRA covering the Off-Project Area considered in this Agreement. The processes for the development, review and approval of such Plans will conform to those specified in the KBRA, to the extent that such processes are not fully described herein.
- **9.10.** Decisions on whether to issue incidental take permits and other Regulatory Assurances under this Agreement will be based on ESA Section 10, USFWS and NMFS' implementing regulations and policy, and best available scientific and commercial data, and shall be coordinated with requirements of sections 3 and 4 of this Agreement to the maximum extent practicable and consistent with ESA Section 10 and USFWS and NMFS implementing regulations and policy.
- **9.11.** By entering into this Agreement, USFWS and NMFS are not prejudging the outcome of any process under the ESA, National Environmental Policy Act (NEPA) or any USFWS and NMFS implementing regulations, and NMFS and USFWS are required to make determinations and take actions as necessary to meet the requirements of the ESA, NEPA, and implementing regulations.
- 9.12. The Non-Federal Parties shall support authorizations and appropriations of funding, in the amount estimated in KBRA, for development of the GCP; actions necessary for review of incidental take permit applications; actions necessary for issuance of incidental take permits; and measures for satisfaction of the incidental take permit issuance criteria that are not funded under other provisions of this Agreement, including measures for minimization and mitigation of incidental take, and including monitoring programs required for incidental take permits. USFWS and NMFS shall coordinate with other Parties to help identify if any other sources of funding are available for the actions described above in this section.

- **9.13.** Parties participating in the development of the GCP(s) or any HCP(s) shall attempt to resolve any disputes pursuant to section 11 of this Agreement. The parties responsible for developing the GCP or HCP shall report to the covered parties quarterly or otherwise as necessary regarding any unresolved disputes or delays in development of the GCP(s) or any HCP(s) or delays in subsequent procedures necessary to reach a decision(s) on issuance of the incidental take permit(s) that may result in failure to meet the goal of a decision(s) on whether to issue such permits.
- **9.14.** It is the intent of the Parties that the Riparian Management Agreements will, to the extent practicable, be consistent with applicable provisions of the Oregon Department of Agriculture's Water Quality Management Program and associated rules for the Klamath Basin, considering the objectives of this Agreement.
- **9.15.** Subject to the enactment of Federal Authorizing Legislation and availability of funds, the GCP(s) and/or HCP(s) described in this Agreement will be developed starting no later than three years after the date that the United States becomes a Party to this Agreement under subsection 12.9, with the objective of completion of the GCP(s) and/or HCP(s) within five years.

10. CONTINGENCIES

- 10.1. Secretarial Affirmative Notice. This Agreement will become permanent when the Secretary of the Interior has published an Affirmative Notice in the Federal Register that all of the conditions listed below have been achieved. The Secretary shall publish the Notice within 120 days of the latest-occurring of the events listed below being completed:
 - **10.1.1.** Federal legislation is enacted authorizing Federal participation in the WUP that is materially consistent with this Agreement, or any material inconsistency in such legislation has been resolved pursuant to subsection 12.2.3.3.4.
 - **10.1.2.** Federal legislation is enacted authorizing Federal participation in the Riparian Program that is materially consistent with this Agreement or any material inconsistency in such legislation has been resolved pursuant to subsection 12.2.3.3.4;
 - **10.1.3.** Federal legislation is enacted that authorizes the execution and implementation of the KBRA and KHSA in a manner that is materially consistent with Appendix A to the KBRA;
 - **10.1.4.** Federal legislation is enacted authorizing Federal participation as a voting member of the JME that is materially consistent with this Agreement;
 - **10.1.5.** The State is authorized to participate in the JME as a voting member of the JME;
 - **10.1.6.** Federal funds have been appropriated and made available for implementation of this Agreement in the amounts provided for in section 2 of this Agreement and in the water, fisheries, and tribal sections of the KBRA as established in Appendix C-2 of the KBRA as it may be revised from time to time;
 - **10.1.7.** To the extent necessary, funds in addition to those described in subsection 10.1.6 have been secured from sources other than those described in subsection 10.1.6 for implementation of this Agreement in the amounts necessary for the JME and LE to carry out their functions under sections 3, 4, 5, 7, and 8 of this Agreement.
 - **10.1.8.** The WUP Region Volumes have been met, materially consistent with the WUP in section 3 of this Agreement;

- **10.1.9.** There is Sufficient Participation in the Riparian Program, materially consistent with sections 3 and 4 of this Agreement;
- **10.1.10.** The Klamath Tribes have notified the Secretary of their willingness to proceed with this Agreement following enactment of the Federal Authorizing Legislation in subsection 12.3 or any amendments thereto pursuant to subsection 12.2.3.3.4, either by confirming that such legislation is materially consistent with this Agreement, or by confirming that they are willing to proceed with the Agreement despite the material inconsistency, or by ratifying an amended Agreement. Any such confirmation or ratification will have no effect on the effectiveness of the legislation authorizing the KBRA or KHSA.
- **10.1.11.** The LE has determined, as soon as practicable, following the enactment of the Federal Authorizing Legislation, that the Federal Authorizing Legislation is materially consistent with this Agreement, or, consistent with subsections 8.4.1, 12.1.1, 12.1.2, and 12.2.3.3.4, any necessary amendments to this Agreement have been made and approved by the LE to conform it to the provisions of the Federal Authorizing Legislation. Any such confirmation or ratification will have no effect on the effectiveness of the legislation authorizing the KBRA or KHSA.
- **10.1.12.** OWRD has determined, as soon as practicable, following the enactment of the Federal Authorizing Legislation, that the Federal Authorizing Legislation is materially consistent with this Agreement, or, consistent with subsection 12.1.1, 12.1.2, and 12.2.3.3.4, any necessary amendments to this Agreement have been made and approved by OWRD to conform it to the provisions of the Federal Authorizing Legislation.
- **10.1.13.** The United States has signed this Agreement pursuant to the Federal Authorizing Legislation.
- **10.1.14.** State funds have been secured and expended in an amount sufficient to acquire or otherwise make permanently available sites to provide public access that are suitable for the exercise of the Tribal treaty rights as described in subsection 6.1 of this Agreement.
- **10.1.15.** The Klamath County Circuit Court has issued a Final Decree (or applicable Partial Final Decree if permitted) affirming the Provisionally Settled Tribal Water Right Claims and such Decree is sustained following any and all appeals; or there are no remaining non-Party exceptors to the Provisionally Settled Tribal Water Right Claims, and the Klamath County Circuit Court has issued an order affirming

the Provisionally Settled Tribal Water Right Claims conditional on the withdrawal of exceptions by the Adjudication Parties as described in subsection 1.1 and such order is sustained following any and all appeals; or, (1) the provisions of subsections 10.4.1 to 10.4.3 have been unsuccessful in resolving non-Party exceptions by September 30, 2016, (2) the Oregon Legislature has passed legislation materially consistent with that described in subsection 12.4 of this Agreement, and (3) the Parties have thereafter successfully applied the statute under the procedures identified in subsections 10.4.4 and 10.4.5 that resulted in the Decree's conditional affirmation of the Provisionally Settled Tribal Water Right Claims and dismissed all non-Party exceptions, and such Decree is sustained following any and all appeals.

10.2. Secretarial Negative Notice. If the Secretary determines that one or more of the conditions in subsection 10.1 has not been or cannot be achieved, the Secretary shall inform the Parties of that preliminary determination by letter. Thereafter, any Party may initiate the Meet and Confer procedures of section 11 of this Agreement to seek to take the necessary actions outside of this Agreement or to amend this Agreement to provide a reasonable likelihood that the events in subsection 10.1 will occur. Any such process under section 11 of this Agreement shall be concluded within 12 months of that letter. If, following this process, the Secretary determines that there is no reasonable likelihood that an Affirmative Notice under subsection 10.1 will occur, the Secretary shall publish a Negative Notice in the Federal Register that the Affirmative Notice under subsection 10.1 will not be published.

Upon publication of the Negative Notice under subsection 10.2, this Agreement shall terminate on the date it is published, except that if judicial review of the Negative Notice is Timely sought pursuant to subsection 10.3, this Agreement shall not terminate unless and until the Negative Notice is sustained following any and all appeals. If this Agreement terminates, the Parties may return to litigation in the Klamath County Circuit Court on the Provisionally Settled Tribal Water Right Claims and any additional claims listed in subsection 1.3 that were subsequently resolved, and any exceptions filed against those claims.

10.3 Judicial Review of Secretarial Finding. The Secretary's Affirmative Notice published pursuant to subsections 10.1 or Negative Notice pursuant to subsection 10.2 is deemed a final agency action, reviewable for purposes of 5 U.S.C. §§ 701 to 706.

- 10.4 Common Litigation Strategy Regarding Non-Party Exceptions to the Provisionally Settled Tribal Water Right Claims. For purposes of this section, the Adjudication Parties have a common interest in seeing that the Provisionally Settled Tribal Water Right Claims that are at the root of this Agreement are Timely determined on the terms of the water rights found in the FFOD. To protect these interests, the Adjudication Parties will take the following actions:
 - **10.4.1.** Upon expiration of the deadline for filing exceptions to the Tribal Water Rights in the Klamath County Circuit Court, the Non-Federal Parties and the Secretary shall immediately determine if any person filed an exception to the Provisionally Settled Tribal Water Rights who is not a signatory to this Agreement ("non-Party exceptor"). If none were filed, subsection 10.4 has no further applicability.
 - **10.4.2.** If there are one or more non-Party exceptor(s), the Non-Federal Parties and the Secretary shall immediately begin efforts to engage the exceptor(s) to (1) withdraw or dismiss the exception(s) or (2) become a signatory to this Agreement. These efforts may last no more than 90 days but may last less time if the Non-Federal Parties and the Secretary agree that a good faith attempt at securing either option (1) or (2) has been made. If either option (1) or (2) are successful in eliminating all non-Party exceptions, subsection 10.4 has no further applicability. Otherwise, the Non-Federal Parties and the Secretary shall proceed to take actions under the next subsection. The Non-Federal Parties and the Secretary shall not offer to make any material alteration in this Agreement to induce a non-Party exceptor to either withdraw or dismiss an exception or to become a signatory to this Agreement.
 - **10.4.3.** The Non-Federal Parties and the Secretary shall assess whether any of the exceptions of the non-Party exceptors ("non-Party exceptions") are subject to a good-faith challenge on legal grounds, such as, the exception being filed untimely or the non-Party exceptor lacking standing. The Non-Federal Parties and the Secretary agree to support the necessary filing of documents with the Klamath County Circuit Court to provide a means to make such legal challenges as soon as reasonably possible after the filing of exceptions and completion of negotiations under subsection 10.4.2. If such legal challenges are successful in dismissing all non-Party exceptions and prevailing on any interlocutory appeals, subsection 10.4 has no further applicability. If any such legal challenges are not successful in dismissing all non-Party exceptions, the Non-Federal Parties and the Secretary shall proceed to take action under the next subsection.

- 10.4.4. Within 90 days of the Notice by the Klamath Tribes to the Secretary under subsection 10.1.10, the Adjudication Parties will file a stipulation and proposed order in the Klamath Adjudication that implements the relevant commitments in sections 1, 3, 4 and 10 of this Agreement, that condition the exercise of the Tribal Water Rights and establish the point at which any conditional approval of the Tribal Water Rights is either permanent, or is not and the Adjudication Parties return to litigation. The intended result is an order reflecting the conditions on the exercise of the Tribal Water Rights created by this Agreement and subject to the contingencies in subsection 10.1.
- 10.4.5. This subsection employs the anticipated legislation described in subsection 12.4 and is only applicable if the Oregon Legislature Timely adopts the legislation described therein. Within 60 days of the effective date of that legislation, the Non-Federal Parties and the Secretary shall initiate all reasonable steps to have a Partial Final Decree entered on the order resulting from subsection 10.4.4 that accomplishes the following: conditionally approves the Provisionally Settled Tribal Water Right Claims; rejects and dismisses the non-Party exceptions; and allows all of the exceptions of the signatories to this Agreement to be fully litigated in the event the Secretary publishes a Negative Notice under subsection 10.2 of this Agreement.
- **10.4.6.** If such legislation has not been Timely passed by the Oregon Legislature, or the litigation under subsections 10.4.4 and 10.4.5 has not successfully resolved the non-Party exceptions, the Secretary may initiate the Meet and Confer procedures of section 11.
- 10.5. Court Proceedings Prior to Secretarial Affirmative Notice. The Parties agree that if, notwithstanding the Best Efforts of the Parties, the Klamath County Circuit Court schedules proceedings pertaining to hearing exceptions to the Provisionally Settled Tribal Water Right Claims prior to publication of the Affirmative Notice described in subsection 10.1, except for such proceedings as may occur pursuant to subsections 10.4.2 through 10.4.4, the Parties will initiate the Meet and Confer procedures of section 11.

11. MEET AND CONFER

- 11.1. Procedure. Any Party may request that any other Party(ies) meet in good faith to reach a mutually agreeable resolution of circumstances not foreseen in this Agreement or a dispute over the implementation of this Agreement. Thereafter, a request to Meet and Confer shall be made in writing and shall describe the nature of the unforeseen circumstance(s) or dispute, and the need to resolve the circumstance(s) or dispute. The Parties will meet and confer within a reasonable time to attempt to resolve the circumstance(s) or dispute. If the resolution requires an amendment to this Agreement, the Parties will follow the process in subsection 12.1. If the resolution does not require an amendment to this Agreement, the Parties will document the resolution in writing, and report it to the United States, the Klamath Tribes, the State, the LE and the JME. If the Parties are unable to resolve the circumstances or dispute within a reasonable time, they will request mediation as described in subsection 11.2.
- 11.2. Non-Binding Mediation. If the circumstance(s) or dispute is not resolved through the procedures described in subsection 11.1., the participating Parties shall use a neutral mediator, with the costs of the mediator allocated such that the Party initiating the process under subsection 11.1 will pay at least ten percent of the costs. The participating Parties shall select a mediator within 30 days. The mediation process shall be concluded not later than 60 days after the mediator is selected. The above time periods may be shortened or lengthened upon mutual agreement of the participating Parties. The participating Parties will document the resolution of the mediation in writing, and report it to the United States, the Klamath Tribes, the State, the LE and the JME. Mediation under this subsection is not binding on the Parties to the mediation. If the resolution requires an amendment to this Agreement, the Parties will follow the process in subsection 12.1.

12. GENERAL TERMS

12.1. Amendments.

- 12.1.1. Except as provided in subsections 12.1.2 through 12.1.4, the Adjudication Parties may amend section 1 of this Agreement only by written agreement of the BIA, the Klamath Tribes, the State, and the Consent of the other Adjudication Parties. Prior to March 31, 2015, amendments to other sections of this Agreement may be made by written agreement of the United States, the Klamath Tribes, the State, and the Consent of the other Parties. Following March 31, 2015, amendments to sections of this Agreement other than section 1 may be made by unanimous written agreement of the Klamath Tribes, the United States, the State, and the LE.
- 12.1.2. Amendment may be made to this Agreement only in the circumstances specified in subsections 12.2.3.3.4, and 12.5 through 12.8 below; except, amendments to sections 3 and 4 are not subject to this provision and may be amended as described therein. Further, any amendment must be consistent with the Authorizing Legislation and Applicable Law, or if not, such amendment must be explicitly conditioned on approval in subsequent legislation.
- **12.1.3.** The Upper Klamath Water Users Association, Resource Conservancy, Sprague River Water Resource Foundation, and Fort Klamath Critical Habitat Landowners agree to recommend to their members approval of this Agreement and will use their Best Efforts to obtain the signatures of their members in their individual capacities as Parties to this Agreement.

A person or entity who signs this Agreement within 30 days of the Effective Date will become a Party to this Agreement. Thereafter, a person or entity who wishes to become a Party to this Agreement after the Effective Date may submit a written request to the JME. The JME may approve the addition of the person or entity as a Party by Consensus, if the JME finds that its participation will contribute to the effective implementation of this Agreement. Upon approval of a request, any such subsequent Party shall execute this Agreement.

12.1.4. The LE shall become a Party to this Agreement upon its formation and Effective Date of this Agreement.

12.2. Obligations Under This Agreement.

- **12.2.1. Regulatory Approvals.** The Parties shall support the application for and granting of Regulatory Approvals not inconsistent with the Agreement, subject to subsections 13.2.5 and 13.2.6.
- **12.2.2. Defense of Agreement.** Subject to subsection 13.2., each Party shall support and defend this Agreement in each applicable venue or forum, including any administrative or judicial action in which it participates, and which concerns the validity of any Regulatory Approval or Authorizing Legislation.
- **12.2.2.1. Litigation.** Subject to subsection 12.2.3.3.3, the form of support or defense in such administrative or judicial action shall be left to the discretion of each Party, including what specific litigation positions to recommend or take in any such action, except as set forth in subsections 1.6 and 10.4. This section does not apply to a dispute or action challenging the adequacy of a Party's performance of an obligation under this Agreement.
- 12.2.2. Comments. Each Party may comment on the consistency of any plan, other document, or data arising in the implementation of this Agreement and not otherwise set forth in the Exhibits. The Parties acknowledge that their comments may conflict due to differing good-faith interpretations of the applicable obligations under this Agreement.
- **12.2.3. Scientific Research.** Nothing in this Agreement shall prevent scientific research or the publication of the same by any Party.

12.2.3. Obligation to Implement

- **12.2.3.1. General.** Each Party shall implement each of its obligations under this Agreement in good faith and with due diligence. Any obligation identified as an obligation of all of the Parties does not obligate any individual Party to take any action itself or itself make any specific commitment other than to participate in the applicable procedures.
- 12.2.3.2. Cooperation Among the Parties. The Parties shall cooperate in the implementation of this Agreement. A Party shall not act in a manner that results in an action or requirement that is inconsistent with the Agreement unless necessary to comply with statutory, regulatory or other legal responsibilities; in which event, the Party shall provide Timely Notice to other Parties to permit Meet and Confer procedures pursuant to section 11.

12.2.3.3. Timeliness

- <u>12.2.3.3.1.</u> General. Each Party shall use Best Efforts to Timely implement its obligations.
- 12.2.3.3.2. Cooperation. The obligation to assist in the implementation of this Agreement is joint and several to all Parties, as well as individual to each Party. In seeking funding, or using Meet and Confer procedures or litigation, as described above, each Party will be mindful of the efforts of other Parties and will seek to cooperate to achieve efficiencies and avoid duplication or other unnecessary costs or efforts.
- 12.2.3.3.3. Obligation to Cure. The filing of an action by one Party against another Party over rights or obligations addressed in this Agreement is deemed to constitute a failure of the mutual obligations set forth in this Agreement. Such failure triggers, on the part of all-Parties, an obligation to preserve the benefits of the Agreement for all Parties, including any Party who is, or could be, adversely affected by such litigation. Parties directly affected by the action, as well as other Parties with interests in, or obligations in this Agreement related to, the subject of the action, shall: (1) meet and confer promptly and in good faith to confirm that the Meet and Confer procedures in section 11 have identified the nature of the dispute, the provisions of this Agreement which failed to achieve the bargained-for benefits as to affected Parties, and the potential remedies for the defect; or (2) if not already completed, take appropriate steps to secure Timely performance of obligations under this Agreement; or (3) seek amendment of the Agreement pursuant to subsection 12.1; or (4) otherwise make Best Efforts to develop and implement a functional cure to preserve the bargained-for benefits under the Agreement for all Parties, including the Parties adversely affected by the litigation; and (5) seek a supervised settlement conference in the adjudicatory forum and advise the court or presiding officer of their Contractual Obligations under this Agreement, including this provision of the Agreement. Further, if an action is commenced against a Party by a non-Party that relates to rights or obligations addressed in this Agreement, the Parties shall to the maximum extent practicable and applicable, comply with the obligations of (2) through (5) above.
- 12.2.3.3.4. Authorizing Legislation. Within 60 days after enactment of Authorizing Legislation, any Party which believes that such legislation is not materially consistent with this Agreement shall provide a Meet and Confer Notice pursuant to section 11. "Material consistency," as related to this Agreement means that the legislation

does not substantially diverge from the bargained-for benefits of this Agreement. The purpose of the Meet and Confer procedures in this circumstance shall be to develop a resolution, such as (1) a joint effort to secure an amendment to the legislation or (2) a mutually agreeable amendment to Contractual Obligations under this Agreement, to restore the bargained-for benefits.

- 12.2.3.4. Extension of Time. Except as otherwise provided in this Agreement, if any Party requires more time than permitted by this Agreement to perform an obligation, that Party shall provide Notice to other Parties in a Timely manner upon recognition of the need for the extension, but before the applicable deadline. The Notice shall explain: (1) the obligation that the Party is attempting to perform; (2) the reason that performance is or may be delayed; (3) the steps the Party has taken or proposes to take to Timely complete performance; and (4) the Party's request for additional time to complete performance. If any other Party disputes the request for additional time, that other Party shall initiate the Meet and Confer stated in section 11. This provision does not apply to any applicable deadline imposed by Applicable Law.
- **12.2.3.5. Environmental Review.** Each Federal agency shall undertake environmental review as required by Applicable Law in the development of, and before commitment to, any program, plan, policy, or implementing action provided for under this Agreement.
- 12.3. Support for Necessary Federal Authorizing Legislation. The Non-Federal Parties agree to support in the Federal Authorizing Legislation provisions that would: (1) confirm the commitments made herein by the United States and the Tribes and that the commitments are effective and binding according to their terms; (2) authorize the Tribes to issue the voluntary relinquishment and release of claims against the United States as provided in subsection 2.5; (3) authorize and direct the Secretary to publish the Notice identified in subsection 10.1 or the Negative Notice identified in subsection 10.2, as applicable; (4) provide for judicial review of a decision by the Secretary pursuant to subsection 10.1 or subsection 10.2 under the Administrative Procedure Act, 5 U.S.C. §§ 701-706, except that the time period in which to commence such action shall be limited to one year; (5) confirm that this Agreement and State legislation that is materially consistent with subsection 12.4 is consistent with 43 U.S.C. § 666; (6) authorize the BIA and the Tribes to enter into further settlement agreements resolving any of the claims identified in subsection 1.3.1. on terms that are consistent with the governing law and the totality of the circumstances; and, (7) provide that operations of the JME and JME Technical Team shall not be subject to the Federal Advisory Committee Act, 5 U.S.C. App.

- **12.4. Support for Necessary Oregon Authorizing Legislation.** The Parties agree to support Oregon legislation that would provide authority to the Klamath County Circuit Court to hear exceptions on less than all of the water right claims in the Adjudication and issue more than one final and appealable decree ("partial final decrees"). The Oregon legislation will allow for partial final decrees in the following circumstances:
 - **12.4.1.** Issuance of a partial final decree for the Provisionally Settled Tribal Water Right Claims after a stay of all proceedings for all these claims until either the Agreement becomes permanent or if this Agreement is terminated by operation of section 10, then issuance of a partial final decree for the Tribal Water Right Claims;
 - **12.4.2.** Issuance of a conditional partial final decree that will obviate the need for parties in the Klamath Adjudication to have to undertake the actions specified in KBRA Sections 15.3.2.B.ii.b(2) (8) to achieve the same result;
 - 12.4.3. If necessary under the circumstances described in subsection 10.4.5., issuance of a conditional partial final decree conditionally determining the Provisionally Settled Tribal Water Right Claims and any claims listed in subsection 1.3 that are subsequently resolved, but allowing the exceptions to these claims to be revived and litigation on those claims to be resumed if the Agreement is terminated pursuant to section 10;
 - **12.4.4.** Issuance of a partial final decree conditionally approving the following:
 - **12.4.4.1**. the conditional withdrawal by the Tribes and the BIA of the Off-Reservation Claims in subsection 1.1 and those Off-Reservation Claims that have been subsequently resolved pursuant to subsection 1.3.1;
 - **12.4.4.2.** OWRD's conditional withdrawal of the findings of fact and conclusions of law pertaining to the merits of the withdrawn Off-Reservation Claims, the merits of any withdrawn parts of partially withdrawn claims, and any other language in the FFOD addressing Federal Indian reserved water rights to off-reservation water; and
 - **12.4.4.3.** In the event the Agreement is terminated under section 10 of the Agreement, the Court will enter an order providing that: (1) the conditional withdrawal of the Off-Reservation Claims is rescinded; (2) the exceptions to the Off-Reservation Claims are revived; (3) the

findings of fact and conclusions of law related to the denial of the Off-Reservation Claims in the FFOD and any other language in the FFOD addressing Federal Indian reserved water rights to off-reservation water are re-entered; and (4) the litigation on the Off-Reservation Claims is resumed; and

12.4.5. Issuance of a single partial final decree determining water right claims not subject to the subsections above;

Provided, that any partial final decree issued subsequent to a conditional partial final decree pursuant to 12.4.3 or 12.4.4, as a result of the revival of exceptions and additional litigation, may not be adverse to the exceptors as compared to the determination in the conditional partial final decree; and *further provided*, that a conditional partial final decree does not constitute a decreed water right for the purposes of Oregon water law until such decree is no longer conditional.

12.5. Force Majeure.

- **12.5.1. Suspension of Obligation.** During a Force Majeure event, and except as otherwise provided in this Agreement, the affected Party shall be relieved of any specific obligation directly precluded by the event, as well as those other obligations whose performance is materially impaired, but only for the duration of such event.
- **12.5.2. Remedies.** If a Force Majeure event occurs the following procedure will be followed. The affected responsible Party shall provide Notice within three days of the onset of the event. Such Notice shall describe the occurrence, nature and expected duration of such event. That initial Notice shall be followed by further Notice within seven days of the onset of the event, describing the steps the Party has taken or proposes to be taken to prevent or minimize the interference with the performance of any affected obligation under this Agreement. The responsible Party shall thereafter provide periodic Notice to the other Parties of the efforts to address and resolve a Force Majeure event. If any other Party disputes the responsible Party's claim of a Force Majeure event, or the adequacy of the efforts to address and resolve such event, such Party shall initiate the Meet and Confer procedures stated in section 11 and shall seek a modification of the schedule or other element of the obligation to preserve the bargained-for benefits of this Agreement.
- **12.6. Untimely or Inadequate Funding.** If, notwithstanding Best Efforts in seeking the funding, the Parties do not secure adequate funding on a Timely basis to perform a particular obligation, the Parties shall seek to

agree to an alternative schedule and other appropriate remedies to permit the performance of that particular obligation.

- 12.7. Severability. This Agreement is made on the understanding that each provision is a necessary part of the entire Agreement. However, if any provision of this Agreement is held to be invalid, illegal, or unenforceable by a regulatory agency or a court of competent jurisdiction: (1) the validity, legality and enforceability of the remaining provisions of this Agreement are not affected or impaired in any way; and (2) the Parties shall negotiate in good faith in an attempt to agree to another provision (instead of the provision held to be invalid, illegal, or unenforceable) that is valid, legal, and enforceable and carries out the Parties' intention to the greatest lawful extent under this Agreement.
- **12.8. Changed Circumstances.** If a Party believes that any event subsequent to the Effective Date impairs or threatens to impair the bargained-for benefits, the Parties shall consider whether to amend the Agreement pursuant to the Meet and Confer provisions of section 11.
- **12.9. Federal Agencies as Parties.** Prior to any Federal agency becoming a Party to this Agreement as described below, whenever this Agreement attributes an action to a Federal agency, that attribution states an expectation of the Non-Federal Parties, rather than an obligation of the Federal agency under this Agreement. Prior to the enactment of Authorizing Legislation, neither the United States nor any of its agencies, officers, or employees shall be a Party to this Agreement, or shall be required to implement any obligation under this Agreement. The Non-Federal Parties execute the Agreement having received separate letters from the Department of the Interior and the National Marine Fisheries Service, expressing their intent to take actions consistent with this Agreement to the extent such actions are consistent with the agency's existing legal authorities and appropriations are available for such purposes. Upon enactment of Authorizing Legislation that authorizes and directs Federal agencies to become parties to this Agreement, the following agencies of the United States ("Federal Agency Parties") shall become Parties to this Agreement without any further action or approval:

United States Department of the Interior acting through the Secretary of the Interior, Bureau of Indian Affairs, U.S. Fish and Wildlife Service, and U.S. Geological Survey, and

United States Department of Commerce's National Marine Fisheries Service.

- **12.10. Enforcement.** This subsection 12.10 is subject to subsection 13.2.8.
 - **12.10.1 Contractual Obligations.** A Party may bring an action to enforce any Contractual Obligation under this Agreement; *provided* that no Party may bring an action against the United States or its agencies concerning this Agreement except as provided in subsection 13.2.8 below.
 - **12.10.1.1. Meet and Confer.** A Party may seek to enforce a Contractual Obligation only after compliance with the Meet and Confer procedures in section 11 and the dispute resolution procedures in subsection 7.4 as applicable.
 - **12.10.1.2. Remedy.** In such an action, a disputing Party may only seek specific performance of the Contractual Obligation, or declaratory or other equitable relief, to the maximum extent permitted by Applicable Law. This Agreement does not establish a right to seek relief, or jurisdiction for such relief, against a Party if such relief or jurisdiction does not otherwise exist under Applicable Law.
 - **12.10.1.3. Venue.** The venue for an action to enforce a Contractual Obligation shall be as provided under Applicable Law for obligations of the type of the disputed Contractual Obligation.
 - **12.10.2. Regulatory Obligations.** A Party may bring an action to enforce any Regulatory Obligation, once approved as proposed under this Agreement, only to the extent otherwise provided by Applicable Law. Nothing in this Agreement establishes a right to enforce a Regulatory Obligation, or jurisdiction for such enforcement, against a Party if such right or jurisdiction does not otherwise exist under Applicable Law.
 - **12.10.2.1. Dispute Resolution.** A Party may seek to enforce a Regulatory Obligation against another Party, only after compliance with the Meet and Confer procedures in section 11.
 - **12.10.2.2. Remedy.** In such action, a disputing Party may seek whatever remedies are ordinarily available for enforcement of obligations of the type of the disputed Regulatory Obligation. This Agreement does not establish any special remedy for such enforcement.
 - **12.10.2.3. Venue.** The venue to enforce a Regulatory Obligation shall be as provided under Applicable Law.

- **12.10.3. Planning Obligations.** Consistent with subsections 13.2.6 and 12.2.3.5, no Party may assert that a Federal or State agency's obligation for a plan or policy under this Agreement is a pre-decisional commitment to any action.
- **12.10.4. No Third Party Beneficiaries.** This Agreement does not create any right in the public, or any member thereof, as a non-Party beneficiary. The rights and obligations of the Parties with respect to non-Parties shall remain under Applicable Law.
- 12.10.5. Actions Against a Federal, State, or Tribal Party. A Party may bring an action against the United States or its agencies, the State, or the Klamath Tribes only to the extent and in the manner provided by Applicable Law. Nothing in this Agreement establishes any jurisdiction or remedy against the United States or its agencies, the State, or the Klamath Tribes if such jurisdiction or remedy does not otherwise exist under Applicable Law. A Party may bring an action against the United States or its agencies, the State, or the Klamath Tribes only after compliance with the dispute resolution procedures in Section 7.4, and the Meet and Confer procedures in Section 11 and subject to Section 13.2.8.
- **12.11. Successors and Assigns.** This Agreement shall be binding on and inure to the benefit of the Parties and their successors and assigns, unless otherwise specified in this Agreement.
- **12.12. Joint Venture.** Except as expressly provided, this Agreement does not and shall not be deemed to make any Party the agent for, partner of, or joint venture with, any other Party.

12.13. Governing Law.

- **12.13.1. Contractual Obligation.** A Party's performance of a Contractual Obligation arising under this Agreement shall be governed by (1) applicable provisions of this Agreement and (2) Applicable Law for obligations of that type.
- **12.13.2. Regulatory Obligation.** A Party's performance of a Regulatory Obligation, once approved as proposed by this Agreement, shall be governed by Applicable Law for obligations of that type.
- **12.13.3. Reference to Statutes or Regulation.** Any reference in this Agreement to any Applicable Law shall be deemed to be a reference to a statute or regulation, or successor, in existence as of the date of the action in question.

- **12.14. Elected Officials Not to Benefit.** This Agreement shall not provide any benefit for any elected official, other than the benefits provided to all Parties.
- 12.15. Entire Understanding. The Parties to this Agreement intend that this Agreement is consistent with Sections 15.3.5, 16, and 22.2, and the revised Appendix C-2 of the KBRA. Section 15.3.5 of the KBRA is attached as Exhibit M. This Agreement constitutes the final, complete and exclusive agreement and understanding among the Parties with respect to the subject matter of this Agreement. Other than the Exhibits to this Agreement, which are attached hereto and incorporated throughout this Agreement by reference, no other document, representation, agreement, understanding or promise, constitutes any part of this Agreement.
- 12.16. Treatment of Communications Related to Agreement. To the fullest extent allowed by Applicable Law all documents and communications related to the development, execution, or submittal of this Agreement to any agency, court, or other entity, shall not be used as evidence, admission, or argument in any forum or proceeding for any purpose. This provision does not apply to the results of studies or other technical information developed for use by the United States or its agencies, the State, or the Klamath Tribes. This provision does not apply to any information that was in the public domain prior to the development of this Agreement or that became part of the public domain at some later time through no unauthorized act or omission by any Party. In the event of the Negative Notice under subsection 10.2, all Parties shall continue to maintain the confidentiality of all settlement communications.

This provision does not prohibit the disclosure of: (1) any information held by a Federal agency that is not protected from disclosure pursuant to the Freedom of Information Act or other Applicable Law; or (2) any information held by a state or local agency that is not protected from disclosure pursuant to the Oregon Public Records Law, or (3) under Applicable Law.

13. LEGAL RESPONSIBILITIES, RESERVATION OF RIGHTS, AND PRECEDENTS.

13.1. Compliance with Legal Responsibilities. Except as otherwise expressly provided, by executing this Agreement, each Party represents that it believes that this Agreement is consistent with its statutory, regulatory, or other legal obligations for conservation, use, or management of affected resources of the Upper Klamath Basin. In the implementation of this Agreement, the United States and its agencies, the State, and other Parties as applicable, shall comply with all applicable legal authorities, including Authorizing Legislation, National Environmental Policy Act, Endangered Species Act, Clean Water Act, National Historical Preservation Act, Native American Graves Protection and Repatriation Act, and other Applicable Law.

13.2. Reservation of Rights.

- 13.2.1. Generally. Nothing in this Agreement is intended or shall be construed to affect or limit the authority or obligation of any Party to fulfill its constitutional, statutory, and regulatory responsibilities or comply with any judicial decision. Nothing in this Agreement shall be interpreted to require the United States or its agencies, the State, or the Klamath Tribes to implement any action which is not authorized by Applicable Law or where sufficient funds have not been appropriated for that purpose by Congress or the State. The Parties expressly reserve all rights not granted, recognized, or relinquished in this Agreement.
- 13.2.2. Reservations Regarding Federal Appropriations. All actions required of the United States or its agencies in implementing this Agreement are subject to appropriations by Congress. Nothing in this Agreement shall be interpreted as or constitute a commitment or requirement that the United States or its agencies obligate or pay funds in violation of the Anti-Deficiency Act, 31 U.S.C. § 1341, or other Applicable Law. Nothing in this Agreement is intended or shall be construed to commit a Federal official to expend Federal funds not appropriated for that purpose by Congress. Nothing in this Agreement is intended to or shall be construed to require any official of the executive branch to seek or request appropriations from Congress to implement any provision of this Agreement. To the extent that the expenditure or advance of any money or the performance of any obligation of the United States or its agencies, or the Secretary under this Agreement is to be funded by appropriation of funds by Congress, the expenditure, advance, or performance shall be contingent upon the appropriation of funds by Congress that are available for this purpose

and the apportionment of such funds by the Office of Management and Budget. No breach of this Agreement shall result and no liability shall accrue to the United States or its agencies or the Secretary in the event such funds are not appropriated or apportioned.

- 13.2.3. Availability of Public Funds. Funding by the United States, or its agencies, or the State under this Agreement is subject to the requirements of Applicable Law. Nothing in this Agreement is intended or shall be construed to require the obligation, appropriation, reprogramming, or expenditure of any funds by the United States or its agencies, or the State except as otherwise permitted by Applicable Law. Nothing in this Agreement is to be construed as permitting any violation of Article XI, section 7 of the Oregon Constitution or any other law regulating liabilities or monetary obligations of the State of Oregon.
- **13.2.4. Reservations Regarding Legislative Proposals.** Nothing in this Agreement shall be deemed to limit the authority of the executive branch of the United States government to make recommendations to Congress on any particular proposed legislation.
- **13.2.5. Reservations Regarding Regulations.** Nothing in this Agreement is intended or shall be construed to deprive any public official of the authority to revise, amend, or promulgate regulations.
- **13.2.6. No Pre-decisional Commitment.** Nothing in this Agreement is intended or shall be construed to be a pre-decisional commitment of funds or resources by the United States or its agencies, or the State. Nothing in this Agreement is intended or shall be construed to predetermine the outcome of any Regulatory Approval or other action by a the United States or the State and their agencies necessary under Applicable Law in order to implement this Agreement.
- 13.2.7. No Alteration of Environmental Review. Nothing in this Agreement is intended or shall be construed to modify the application of NEPA or other Applicable Law, to the environmental review of any program, plan, policy, or action (or project) under this Agreement. The use of the word, "final," with reference to development or adoption of any program, plan, policy, or action, (1) describes the schedule for such development or adoption and (2) does not modify the application of NEPA or other Applicable Law to such development or adoption.
- **13.2.7.1.** Nothing in this Agreement shall be interpreted to limit the discretion under Applicable Law of the United States or its agencies or the State to alter any program, plan, policy, or action of the United

States or its agencies or the State in response to information and considerations developed during the environmental review process.

- 13.2.7.2. Nothing in this section shall be construed to prevent an action which has independent utility from proceeding before environmental review is complete on any program or plan described in this Agreement, provided that such action itself has been subject to environmental review to the extent required by Applicable Law.
- **13.2.8. Enforceability.** Except as set forth in this section, nothing in this Agreement is intended to be, or shall be construed as, a waiver of sovereign immunity by the United States, the State, or the Klamath Tribes. This Agreement does not obligate the United States or its agencies to affirmatively support this Agreement regarding any State or local legislative, administrative, or judicial action before a state administrative agency or court, except as stated in subsections 1.6 and 10.4.

The Klamath Tribes hereby agrees that if the Klamath Tribes brings or joins litigation in court or an administrative adjudication process, and seeks relief affecting the interests of any other Party under the terms of this Agreement, the Klamath Tribes' participation in the litigation or adjudicative proceeding is deemed a consent to that court's or adjudicatory body's jurisdiction over all defenses and counter-claims, related to this Agreement, in the action. Further, if the Klamath Tribes brings or joins litigation in court or an administrative adjudication process, and seeks relief affecting the interests of any other Party under the terms of this Agreement, it agrees it will not seek to bar such other Parties from asserting or seeking to enforce the provisions of this Agreement by way of defense or counter-claim in such a proceeding. Any counter-claims under the terms of this subsection are limited to a request for equitable remedies and shall not include a request for damages of any type or attorneys' fees.

13.2.9. No Argument, Admission, or Precedent. Nothing in this Agreement or any of the attachments thereto shall be offered for or against a Party, including the United States and its agencies and the State and its agencies, as argument, admission, admission of wrongdoing, liability, or precedent regarding any issue of fact or law in any mediation, arbitration, litigation, or other administrative or legal proceeding, except that this Agreement may be used in any future proceeding to interpret or enforce the terms of this Agreement, consistent with Applicable Law. This Agreement may also be used by any Party, including the United States and its agencies and the State and its agencies, in litigation by or against non-Parties to implement or

defend this Agreement. This section shall survive any termination of this Agreement.

- **13.2.10. Protection of Interests.** Each Party may, in a manner consistent with this Agreement, protect, defend, and discharge its interests and duties in any administrative, regulatory, legislative or judicial proceeding.
- 13.2.11. No Determination of Water Rights by the Agreement. No water rights or water rights claims of any Party are determined or quantified herein. No water rights or potential water rights claims of any non-Party to the Agreement are determined herein. No provision of this Agreement shall be construed as a waiver or release of any tribal water or fishing rights in the Klamath River Basin in California, including claims to such water or fishing rights that have not yet been determined or quantified. The Secretary will not take any action in any proceeding within the Klamath Adjudication that eliminates the existence or quantifies the amount of any tribal water or fishing rights in California.
- 13.2.12. Reservation of the Exercise of the Water Rights of the Klamath Tribes and the BIA. This Agreement does not affect the exercise of any tribal water rights other than the rights that are the subject of the Provisionally Settled Tribal Water Right Claims.
- 13.2.13. State Governmental Functions. Nothing in this Agreement is intended to have the effect of delegating from the State any governmental functions, whether regulatory, proprietary or otherwise, to the JME or to the LE, or to confer on any other party or entity state authority to carry out those governmental functions with respect to any of the programs or activities described in this Agreement.
- 13.3. Calls for Regulation. Nothing in this Agreement is intended to affect the agreement between the Klamath Tribes and the BIA when placing calls pursuant to the May 24, 2013 Protocol Agreement between the Klamath Tribes and the BIA, which is on file with OWRD, amended as needed from time to time.

14. EXECUTION OF THE AGREEMENT.

- **14.1. Authority.** Each signatory to this Agreement certifies that he or she is authorized to execute this Agreement and to legally bind the Party he or she represents. This binding effect applies to all obligations which legally may be performed under existing authorities. This binding effect applies to other obligations arising from new authorities arising pursuant to the Authorizing Legislation.
- **14.2. Counterparts.** This Agreement may be executed in counterparts. Each executed counterpart shall have the same force and effect as an original instrument as if all the signatory Parties to all of the counterparts had signed the same document.
- **14.3. Effective Date.** The Effective Date shall be the date that the signatory Parties listed in section 16, below, have signed this Agreement.

15. DEFINITIONS

As used in this Agreement, the following capitalized terms have the following meanings, throughout this Agreement:

- "Affirmative Notice" means the document published in the Federal Register as described in subsection 10.1.
- "Adjudication Parties" means the State and Parties who are also parties to the Klamath Adjudication.
- "Agreement" means this Upper Klamath Basin Comprehensive Agreement
- "Agricultural Use" means a land use zoning designation of Klamath County for farm or farm and forest use, as specified in ORS chapter 215.
- "Allottee" means a member of the Klamath Tribes who owns an allotment within the former Klamath Reservation and who has a water right based on the Klamath Tribes' Treaty of 1864.
- "Applicable Law" means general law which (1) exists outside of this Agreement, including a Constitution, statute, regulation, court decision, or common law, and (2) applies to obligations or activities of Parties contemplated by this Agreement. The use of this term is not intended to create a Contractual Obligation to comply with any law that would not otherwise apply.
- **"Authorizing Legislation**" means the legislation enacted by Congress and the Oregon legislature to authorize and implement this Agreement. The term "Federal legislation" as used in this Agreement includes but is not limited to Federal Authorizing Legislation.
- "Average Annual Basis" means the average of the annual increase in water volume flowing into Upper Klamath Lake estimated, subject to KBRA Section 16.2.2.F.ii, to result from implementing the Water Use Retirement Program over the hydrologic conditions that existed in the years from 1980-2000, or over a different span of years and hydrologic conditions as determined by the Upper Basin Team and OWRD, and approved by the Klamath Basin Coordinating Council.
- "Baseline Conditions" means the Water Use Retirement Program baseline described in Section 16.2 of the KBRA, as determined pursuant to section 3 of this Agreement.

- **"Best Efforts"** means performance of an obligation or attempt to correct non-performance of an obligation in a reasonable manner and good faith, and with that level and quality of effort appropriate to achieve the goals of this Agreement.
- **"BIA"** means the Bureau of Indian Affairs, an agency within the U.S. Department of the Interior.
- "Call Threshold(s)" means the instream flow threshold associated with a Primary or Secondary SIF Measurement Location, to which the Klamath Tribes and the BIA may call for regulation of junior water rights under the terms of this Agreement.
- **"Consensus"** means that an action is taken by affirmation or unanimous written consent of the voting directors of the Board or, if one or more voting directors object to the action, the Board will follow the dispute resolution procedure set forth in subsection 7.4.
- **"Consent"** means a Party's written affirmation responding to a Notice where such Notice requests the Party's affirmative response, or a Party's failure to respond to such Notice within 30 days of the effective date of such Notice.
- "Contractual Obligations" means those obligations under this Agreement that are not subject to Regulatory Approval.
- **"Effective Date"** means the date that the signatory Parties listed in section 16 have signed this Agreement.
- **"Eligible Landowner"** means an Allottee, and any other person or nongovernmental entity that holds a groundwater irrigation right or a surface water irrigation right in the Off-Project Area.
- "Eligible Riparian Landowner" means an Allottee, and any other person or non-governmental entity that owns land in Klamath County that is: (1) zoned by Klamath County for farm use or mixed farm and forest use (or lands on Larkin Creek described in the definition for Riparian Management Corridor); and (2) irrigated by a surface water right or a groundwater right, with a place of use that includes land within a Riparian Management Corridor.
- **"Extreme Drought"** means that the Governor has issued a drought declaration for an area including the Off-Project Area, and the Natural Resources Conservation Service classifies the drought as extreme.

- **"FFOD"** means the Findings of Fact and Order of Determination in the Klamath Adjudication dated March 7, 2013, as it may from time to time be amended or corrected.
- **"Force Majeure"** means an event beyond the reasonable control of an affected Party that prevents the Timely performance of an obligation despite the exercise of due diligence. Such events may include natural disasters not otherwise addressed in this Agreement as well as all unavoidable legal impediment or prohibitions.
- **"Funding Entities"** include any public or private entity or person providing funding for implementation of this Agreement.
- "Gaining Reach" means a reach of a perennial stream where streamflow is increasing as a result of groundwater discharge to the stream, as shown in Exhibit F.
- **"Green Line"** means the riparian monitoring technique described in the following document: Winward, A. H. 2000. Monitoring the vegetation resources in riparian areas. General Technical Report RMRS-GTR-46. U. S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Ogden, Utah.
- "Initial Net Consumptive Use" means the estimated Net Consumptive Use for a parcel on an average annual basis from 1980-2000, expressed as feet per acre.
- "Irrigation Season" means the period from March 1 to October 31 of every year.
- **"JME Technical Team"** means a group of specialists with appropriate expertise in hydrology, agricultural practices, livestock management, biology, ecosystem restoration, wetland and soil science, or other related disciplines, appointed by the JME and including at least one representative of the LE and the Klamath Tribes, and a representative from each of the United States and Oregon, if those governments wish to participate on the team.
- **"Joint Management Entity" or "JME"** means the entity comprised of the LE, the Klamath Tribes, the United States, and the State that represents the interests of the Parties and that is responsible for overseeing implementation of this Agreement as described in section 7.
- "Landowner Entity" or "LE" means the entity described in section 8.

- **"Klamath Adjudication"** means the proceeding for the determination of surface water rights in the Klamath River Basin, referenced as of the Effective Date of this Agreement in the Klamath County Circuit as Case No. WA 1300001.
- "Klamath Basin Restoration Agreement" or "KBRA" means the agreement dated February 18, 2010, as amended December 29, 2012, and found at www.klamathcouncil.org.
- "Klamath Hydroelectric Settlement Agreement" or "KHSA" means the agreement dated February 18, 2010, and found at www.klamathcouncil.org.
- **"Long-Term SIF Threshold"** means a Call Threshold without a Type A or Type B SIF Adjustment as specified in section 3.
- "Major Decision" means a decision of the JME that, under this Agreement, requires the affirmative vote of all of the voting directors of the Klamath Tribes, the United States, and the LE.
- "Multiple Indicator Monitoring" means the riparian monitoring technique described in the following document: Burton, T. A., S. J. Smith, and E. R. Cowley. 2011. Riparian area management: multiple indicator monitoring (MIM) of stream channels and streamside vegetation. Technical Reference 1737-23, U. S. Department of the Interior, Bureau of Land Management, National Operations Center, Denver, Colorado.
- "Negative Notice" means the document published in the Federal Register as described in subsection 10.2.
- "Net Consumptive Use" means the amount of water consumed (used and transpired by plants) in an area on which irrigation water is applied minus the amount of water that would be lost through evapotranspiration by the same area in the absence of irrigation.
- "NMFS" refers to the United States Department of Commerce's National Marine Fisheries Service.
- "Non-Federal Parties" means the Parties other than the Federal agencies.
- "Notice" means a document contemplated under this Agreement that shall be provided to applicable Parties by electronic mail, unless the sending Party determines that first-class mail or an alternative form of delivery is more appropriate in a given circumstance. A Notice shall be effective upon receipt, but if provided by U.S. Mail, seven days after the date on which it is mailed. For the purpose of Notice, the list of authorized representatives of the Parties

is attached as Exhibit J. The representatives of the United States will be included in the initial list. The JME shall maintain a current distribution list. The Parties agree that failure to provide the JME with current contact information will result in a waiver of that Party's right to Notice under this Agreement. The Party who has waived Notice may prospectively reinstate its right to Notice by providing current contact information to the JME.

"Off-Project Area" means the area by that name shown in Exhibit B.

"Off-Project Restoration Area" means the area by that name shown on Exhibit B.

"Off-Reservation Claims" means those water right claims of the Klamath Tribes and the BIA listed in subsection 1.1 and 1.3.1 that are located entirely outside the boundaries of the former Klamath Reservation and those off-Reservation portions of any such claims that bisect the Reservation boundary.

"OWRD" means the Oregon Water Resources Department.

"Parties" or "Party" means the signatories to this Agreement.

"Post-Implementation Net Consumptive Use" means the estimated Net Consumptive Use for a parcel on an average annual basis from 1980-2000, assuming the applicable WUP Practices had been in effect, expressed as feet per acre.

"Practicably Irrigable Acres" means (for purposes of this Agreement) the place of use of an inchoate water right claim of an Allottee, or the transferee of an Allottee, as determined in the Klamath Adjudication.

"Primary SIF Measurement Location" means the locations shown on Exhibit B, except that if such a location is in the middle of the claim reach of a Tribal Water Right and there is a call by the Tribes and the BIA, then the location is the bottom end of the claim reach.

"Proper Functioning Conditions" means vegetation and soil conditions in a Riparian Management Corridor that are consistent with their long-term potential given stream size, gradient, soil type, elevation, and other related variables. See, for example "Process for Assessing Proper Functioning Condition, BLM Tech. Report 1737-9 1993; Chaney, Elmore and Platts "Managing Change Livestock Grazing on Western Riparian Areas" EPA, July 1993; Chaney, Elmore and Platts, EOA, August 1993.

"Provisionally Settled Tribal Water Right Claims" means the following water right claims of the Klamath Tribes and the BIA in the Klamath Adjudication: 616 and 622 (Upper Klamath Lake); claims 625-630, 634, and 640 and the relevant portion of claim 612 (Williamson River Basin instream flow claims below Klamath Marsh); claims 641-657 and the relevant portion of claim 612 (Sprague River Basin instream flow claims); claims 658-662 and the relevant portion of claim 612 (Sycan River Basin instream flow claims below Sycan Marsh); claims 668-670 and the relevant portion of claim 612 (Wood River Basin instream flow claims), and any claims that are subsequently settled under subsection 1.3.1 of this Agreement.

"Regulatory Approvals" means each permit or other approval under a regulatory statute necessary to implement any of the obligations or activities of Parties as contemplated under this Agreement.

"Regulatory Obligations" means each of those obligations proposed, or activities of Parties contemplated, by this Agreement which are subject to Regulatory Approval and, upon such approval, are enforceable under regulatory authority.

"Retired Riparian Length" has the meaning described in subsection 3.10.2.

"Riparian Guidelines" means the guidelines described in Exhibit H.

"Riparian Management Agreement" means an agreement between the LE and an Eligible Riparian Landowner for restoration and/or management of lands within a Riparian Management Corridor that meets the standards and guidelines developed as described in section 4 of this Agreement.

"Riparian Management Area" means the lands within a Riparian Management Corridor that are specified in a particular Riparian Management Agreement.

"Riparian Management Corridor" means the lands within the Off-Project Restoration Area that: (1) are adjacent to a perennial stream in the Westside Restoration Area, or to a perennial stream in the Off-Project Restoration Area that is subject to a Specified Instream Flow; (2) are zoned for Agricultural Use; (3) are within the place of use of a water right for irrigation; and (4) contain, or would contain in a restored condition, vegetation that is strongly influenced by the presence of stream water. Notwithstanding the foregoing, on Larkin Creek upstream of a point 1.22 miles upstream of its confluence with the Williamson River, parcels that include Larkin Creek and are owned by landowners irrigating land with water diverted from Larkin Creek, are considered to be included in the Riparian Management Corridor for the

purposes of Riparian Management Agreements and determinations of Sufficient Participation for the Larkin Creek SIF Region.

"Riparian Program" means the Riparian Program described in section 4 of this Agreement.

"RMA Ledger" means the ledger described in sections 3 and 4 of this Agreement for purposes of tracking Sufficient Participation, Riparian Management Agreement compliance, and Retired Riparian Length.

"Secondary SIF Measurement Location" means the locations shown on Exhibit B, except that if such a location is in the middle of the claim reach of a Tribal Water Right and there is a call by the Tribes and the BIA, then the location is the bottom end of the claim reach.

"Secretary" means the Secretary of the U.S. Department of the Interior.

"SIF Measurement Locations" mean the points in the Off-Project Restoration Area where Specified Instream Flows are measured for purposes of compliance with the obligations described in sections 3 and 6 of this Agreement.

"SIF Region" means the area irrigated from a diversion or point of appropriation located upstream of a SIF Measurement Location, bounded by any other upstream SIF Measurement Location, within which the Parties agree to achieve and maintain Sufficient Participation in the Riparian Program, and to maintain instream flows located at or above applicable SIF Measurement Locations. In some cases, more than one SIF Region may be contained within a single WUP Region.

"State" refers to the State of Oregon, acting by and through its Water Resources Department.

"Sufficient Participation" means that the minimum length of the Riparian Management Corridor specified in subsection 4.8 of this Agreement is included in Riparian Management Agreements, and that the Eligible Riparian Landowners are in material compliance with the terms of such Riparian Management Agreements.

"Timely" means performance of an obligation or act by the deadline established in the applicable provision, and otherwise in a manner reasonably calculated to achieve the bargained-for benefits of the Agreement.

"Total WUP Volume" means the increase in the total volume of inflow into Upper Klamath Lake over Baseline Conditions by 30,000 acre-feet on an

Average Annual Basis that is to be allocated among WUP Regions and produced by decreasing the Net Consumptive Use of water under this Agreement.

- **"Transition Period"** means the period beginning on the Effective Date and ending on March 31, 2019.
- "Tribal Water Rights" for purposes of this Agreement means only the water rights jointly held by the Klamath Tribes and the BIA, in its capacity as trustee of the Klamath Tribes, in the Klamath Adjudication as those water rights listed in subsection 1.1 and those in subsection 1.3.1 that are subsequently settled, as they were determined in the FFOD, and as later determined in the Klamath Adjudication. In addition "Tribal Water Right" (singular) means one of the water rights described above in this paragraph.
- "Tribal Water Rights Claims" (for the purpose of this Agreement) means the non-consumptive water rights claims filed by the Klamath Tribes and the BIA in the Klamath Adjudication numbered as claims 612, 614-616 and 622-673.
- **"Type A SIF Adjustment"** means the adjustment described in subsection 3.9 and Exhibit E.
- **"Type B SIF Adjustment"** means the adjustment described in subsection 3.10 and Exhibit E.
- "United States" refers to the United States Departments of the Interior, acting through the Secretary of the Interior (Secretary), Bureau of Indian Affairs (BIA), U.S. Fish and Wildlife Service (USFWS), U.S. Geological Survey USGS), and the Department of Commerce's National Marine Fisheries Service.
- "USFWS" means the U.S. Fish and Wildlife Service, an agency within the Department of the Interior.
- "USGS" means the U.S. Geological Survey, an agency within the Department of the Interior.
- "Walton rights" means water rights of a successor to an allottee who satisfies the criteria found in Colville Confederated Tribes v. Walton, 460 F. Supp. 1320 (E.D. Wash. 1978); Colville Confederated Tribes v. Walton, 647 F.2d 42 (9th Cir. 1981); Colville Confederated Tribes v. Walton, 752 F.2d 397 (9th Cir. 1985).
- **"Water Use Agreements"** means agreements entered into between the Landowner Entity and an Eligible Landowner to carry out a Water Use Practice.

"Water Use Development" means actions that develop the beneficial use of water under a water right claim determined in the Klamath Adjudication, including but not limited to claims for Practicably Irrigable Acres by Allottees, and actions to develop a water right permit issued by the OWRD but not fully developed as of December 31, 2001, but not including the development of consumptive uses of water by the Klamath Tribes or by Klamath Tribal members (other than PIA).

"Westside Restoration Area" means the area by that name shown in Exhibit B. The Westside Restoration Area includes Sevenmile, Fourmile, Crane, and Cherry Creeks. The Westside Restoration Area is not in a SIF Region, and there is no applicable Long-Term SIF Threshold or Call Threshold, but this area is in the Wood Valley WUP Region for purposes of that WUP Region Volume. The Westside Restoration Area is eligible for the Riparian Program and for Regulatory Assurances under section 9.

"WUP" means the Water Use Program described in section 3 of this Agreement.

"WUP Ledger" means the ledger used to track the WUP Region Volumes over time to determine ongoing compliance, as described in subsection 3.7.

"WUP Practices" means an action to reduce the Net Consumptive Use of water that is permanent or of a limited term but assured through an agreement with a term of at least one year. WUP Practices include, but are not limited to, Water Use Agreements and other items listed in subsections 3.12 and 3.13.

"WUP Region" means an area within the Off-Project Area that is required to produce a specific portion of the WUP Volume. A WUP Region may contain more than one SIF Region.

"WUP Region Call" means a call based on a shortfall in a WUP Region Volume as shown in the WUP Ledger, as described in Section 3.8.2.

"WUP Region Volume" means the specific portion of the WUP Volume that is to be produced by a specific WUP Region as specified in subsection 3.3.

16. SIGNATORY PARTIES

by high

Signed this 🛭	day of April	2014 by	Governor	John	Kitzhaber,	State of	Oregon.

Signed this 18 day of April 2014 by Don Gentry, Chairman, The Klamath Tribes.



Signed this __ day of April 2014 by Phil Ward, Director, Oregon Water Resources Department.



Signed this 6 day of April 2014 by Garrett Roseberry, Resource Conservancy, and Sprague River Water Resource Foundation.



Signed this Sday of April 2014 by Roger Nicholson, Fort Klamath Critical Habitat Landowners.



Signed	this day of April 2014 by Matthew Walter Water Users Association.	r, Upper Klamath Basin
Signed	l this day of April 2014 by Becky Hyde, Up Association.	pper Klamath Water User
		*
	this day of April 2014 by Cheri Little, Upp Association.	per Klamath Water Users
3		
Signed	this day of April 2014 by Allen Foremen	

		-
Signed	this // day of April 2014 by Kevin Newman.	
<i>u</i> -	Zin non	

t.

Signed this day of April 2014 by Matthew Walter, Users Association.	Jpper Klamath Basin
	n A - b 2
Signed this day of April 2014 by Becky Hyde, Upper Association.	er Klamath Water Users
Ben Hyde	· <u>F</u>
Signed this day of April 2014 by Cheri Little, Upper Association.	Klamath Water Users
Association. B. Jettle	
Signed this day of April 2014 by Allen Foremen	g.
A Company of the second	
Signed this day of April 2014 by Kevin Newman.	

Signed	this day of April 2014 by Larry Nicholson.	
Signed	this day of April 2014 by Linda Long.	
Signed	this day of April 2014 by Melissa Hess.	
Signed	this day of April 2014 by Randall Kizer.	
Signed	this day of April 2014 by Todd Mathis, Ma	this Family Trust.
Signed	this day of April 2014 by Tom Burns.	

PROPOSED UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT March 4, 2014

Signed	this day of April 2014 by Larry Nicholson.	
Signed	this day of April 2014 by Linda Long.	
Signed	this day of April 2014 by Melissa Hess.	
Signed	this day of April 2014 by Randall Kizer.	
	this day of April 2014 by Todd Mathis, Mat	his Family Trust.
Signed	this day of April 2014 by Tom Burns.	

and the second of the second o

Signed this day of April 2014 by Matthew Walter, Upper Klamath Basin Water Users Association.
Matcher Matte
Signed this day of April 2014 by Becky Hyde, Upper Klamath Water User Association.
Signed this day of April 2014 by Cheri Little, Upper Klamath Water Users Association.
Signed this day of April 2014 by Allen Foremen
Signed this day of April 2014 by Kevin Newman.

Signed this day of April 2014 by Larry Nicholson.
Signed this /8 day of April 2014 by Linda Long.
Signed this /2 day of April 2014 by Linda Long.
Janeta Jun
Signed this //day of April 2014 by Melissa Hess.
Alelerea Hess
Signed this day of April 2014 by Randall Kizer.
E = = = = = = = = = = = = = = = = = = =
Signed this day of April 2014 by Todd Mathis, Mathis Family Trust.
Signed this day of April 2014 by Tom Burns.
e i

er e e e e e e e e e e e e e e e e e e	
8 ⁰ 3 ^E 8	
	м 8
Signed this day of April 2014 by Becky Hyde, Up Association.	per Klamath Water User
	# ## ## ## ## ## ## ## ## ## ## ## ## #
	*
Signed this day of April 2014 by Cheri Little, Upp Association.	oer Klamath Water Users
Signed this 🌡 day of April 2014 by Allen Foremen	
Signed this day of April 2014 by Allen Foremen	
Signed this day of April 2014 by Allen Foremen Signed this _ day of April 2014 by Kevin Newman.	

Signed	this			il 2014							
Signed	this	day	of Apr	il 2014	· by	Linda	Long.				
Signed	this							5.			
Signed		day	of Apr		· by	Randa	ul Kize				
Signed		day		ril 2014					this I	Family	7 Trust.
Signed	this	day	of Apr	il 2014	· by	Tom E	Burns.				

Signed th	nis day of April 2014 by Larry Nicholson.
Signed th	nis day of April 2014 by Linda Long.
Signed th	nis day of April 2014 by Melissa Hess.
Signed th	nis day of April 2014 by Randall Kizer.
Signed th	nis day of April 2014 by Todd Mathis, Mathis Family Trust.
Signed th	nis & day of April 2014 by Tom Burns.

THIS PAGE IS INTENTIONALLY LEFT BLANK

17. Adjudication Parties (continued)

Signature State	Date: 3-10-14
Print name	***
Syrce Richardson	Date: 3-12-2014
Print name D. K. Patton DVM Print name	Date: 3 -/2 - 2014
Signature Dave THOMPSON Framoso CATTLE Co.	Date: 3-12-2014
Print name JOHN AMARA SEKS Print name	Date: 3/12/2014

17. Adjudication Parties (continued)

Shree Owens	Date: 4-1-14
SHEREE OWENS Print name PHR; Jack Owens Rand	ches
Signature	Date: 4-1-14
RENNETH OWENS Print name PTR: Jack Oweno Ra	incheo
Candace Owens	Date: 4-1-14
Print name PTR Jack Owers Ranch	? 5
Signature Own	Date: 4//////
Tohn B. Owens Print name PTR! Juck Gowens RA	/
	Date:
Signature	Batte.
Print name	

PROPOSED UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT March 4, 2014

17. ADJUDICATION PARTIES (continued)

willbatk	Date: 4-2-(4)
Signature	,
Steven Mark de Vries	_
Print name	
	Date:
Signature	
Print name	
	Date:
Signature	
Print name	
	Date:
Signature	
Print name	
	· .
•	
	Date:
Signature	
Print name	

PROPOSED UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT March 4, 2014

17. ADJUDICATION PARTIES

Signature Robinson Best LLC	Date: 3-25-14
Print name Padney R Cornelsen 2002 Tru Print name	Date: 3/31/14
Signature	Date:
Print name Signature	Date:
Print name	Date:
Signature Print name	Daw.

Signature Ohenchain	Date: 3/24/14
Print name	
Margard Own Jacobs Signature	Date: 3-24-14
Margaret ANN JACOBS Print name	
Signature Kenneth Wayne Jacobs Print name	Date: 3-24-14
Margas Dacubo Signature Margaret away Jacobs Print name	Date: 3-24-14
Signature ·	Date:
Print name	

17. ADJUDICATION PARTIES

19. O web 5 lyn	Date:	3/13/14
Signature D. Mark Flan Flan Print name	w Brothers	,
Signature E. Ahyrry Signature E. Flygg Print name	Date: 3/15	/14
Print name E FLY99	-	
Signature	_ Date:	. :
Print name	-	
Signature	Date:	
Print name		
Signature	Date:	
Print name		

Difflyin	Date: 3-/3-/4
Signature V DAN MY MI / lignay Print name	
Signature Colin	Date: 3-13-14
Print name	
Signature Signature Print name	Date: 3-13-14
Signature PopSon Print name	Date: 3/13/14
Signature Signature Deckie Bentley Print name	Date: 3/13/14
J+T Ranch Company	

Signature/ Signature/ Print name	Date: 3-12-2014
Manley Susan Bley, Sui Signature Must of The facel ble Print name	The Date: 3/12/2014 Ly Mallere Suss Gles
Signature Samuel Such of Sola Paul Bles Print name	Date: 3/12/2014 V Naclese Suran Bles
Matthew D. Walter Print name	Date: 3/12/14
Signature Richard E. Siemens Print name	Date: 3/13/14

Signature Ken Willowd Print name	Date: March /7/2014
Signature Mike WAnpler Print name	Date: 3-13-19
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

17. Adjudication Parties	
Signature Print name	Date: 10 Mar 14
Signature	Date: 3-16-14
Print name Signature	Date: 3-10-14
Signature DUARTE Livestock/ Print name ERIC DUARTE Signature Chad C. Rabe	Date: 3-10-14
Print name Dick de Vries Print name	Date: 3/11/14

Signature	Date: 3/14/2014
Tom biffith by Matt Giffith P.O.A Print name	
Signature LARRY W NICHOLSON Print name	Date: 3/14/2014
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
rint name	

17. ADJUDICATION PARTIES (continued)	
Signature	Date: 4/14/14
MARTIN NICHOLSON Print name	
Signature	Date: 4/14/14
MARTIN MICHOLSON Groose Nest Ranches, Print name	L.L.C.
Signature	Date:
Print name	
Signature	Date:
Print name	•
Signature	Date:
Print name	

17. Adjudication Parties (continued)	A .
Signature Duant Martin TR Print name	Date: 4/4/
Ocone mortin Sep Signature Duane Martin Ranches LP Print name	Date: 4/11/14
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

Signature Jan Hayla Rand	4-10-,4
Print name	
- 117	•
Signature	Date:
Oignature	
Print name	
	Date:
Signature	
Print name	·
	e.
Signature	Date:
Print name	,
Signature	Date:

Signature	Date: 4/1/14
Ken Tuttle 1 Print name	For Pouble K Ranch
Koren Little	Date: 4/1/14
Signature Karen L. Tuttle Print name	Double K Ranch
Signature	Date:
Print name	
Signature	Date:
Print name	
lighature	Date:
rint name	

17. ADJUDICATION PARTIES

Mary Con Michel TTE Signature NICHOLSON LOVING TRUST MARY ANN NICHOLSON Print name	Date April 6 2014
Signature	
NICHOLSON LOVING LEUS!	
MARY ANN NICHOLSON	
Print name	
	Date:
Signature	
•	
Print name	
•	
	Data
Signature	Date:
Print name	
	Date:
Signature	
Print name	
	•
Signature	Date:
Print name	

17. ADJUDICATION PARTIES

Legald H. Hawkind	Date: 4/3/14
GERALD H. HAWKINS Print name	
Harlowe Ranch LLC Ly Leval H. Howkins, Manager Signature HARLOWE RANCH LLC BY GERALD H. HAWKINS, MANAGER Print name	Date: 4/3/14
Hawkins Cattle Company:	Date: 4/3/14
GERALD H. HAWKINS Print name Elizabeth H. Thave kills ELIZABETH H.HAMKINS Signature Print name	Date: 4/4/14
Signature VILLIAM E. HAWKINS Print name	Date: 04/07/2014

Owens + Hawkins:	
Lorald Hawkins Signature	Date: 4/3/14
GFRALD H. HAWKINS Print name	
Elezabeth N. Hawkins Signature	Date: 4/4/14
Print name	·
Barbara Haukins Letsch Signature	Date: 4/4/14
Barbara Haw Kins Fetsch Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

Signature Signature Print name Print name	Date: 4/15/20/4
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
D.:	

	Date:
Signature	
Print name	
Signature Glen Crouch for Glenk Cone Print name and Shorking J. Cro	
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

Signature Newbordson	_ Date: <u>April 15, 20</u> 14
Mbcc CCC Print name	
Signature Leihelan	Date: <u>Opril 15, 2014</u>
Clare-water CC C	<u>.</u>
Magn Kullulwon Signature	Date: 9728 15, 2014
ROUER WICHULSON RI Print name	ANCHES
May beckelin	Date: <u>April 15, 2014</u>
ORONOTIUE TIMBERYAND Print name	SILC
Signature	Date:
Print name	

Signature	_ Date: April 15, 2019
WOOD RIVER DISTRICT Print name	IMP.CO
Signature	_ Date:
Print name	<u>-</u>
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

William & oftill	Date: 4//16/14
Signature	
Print name: Lillin m [Ji]	
	Date:
Signature	·
Print name	<u></u>
	Date:
Signature	
Print name	
	Date:
Signature	
Print name	
	Date:
Signature	
Drint name	·

Es ash Mantine () Signature	Kem	Date: 4-15-20
Elaine G. Kes Print name	ns 1992 Tru	st
Signature		Date:
D.:. 4		,
Print name		
Signature		Date:
Print name		
		,
Signature		Date:
	,	
Print name		
Signature		Date:
_		

17. ADJUDICATION PARTIES (continued	
RIUBRS PRINGS RAWCI	
Sienes	
Signature	Date: 4-30-14
PIERRE A. VERN	
Print name	
Ryger SPRINGS RANCH	
N I i	1/3
Signature	Date: 4-30-14
PICARD B. KERN	
Print name	·
RIVER SPRINGS RANCH	
AND G	Date: 4-30-14
Signature	Date: 1 - 50 / 4
~ 1.24	
BAD FORD M. KORN Print name	
rim name	
	Data
Signature	Date:
Print name	
Tare Herale	
•	Date:
Signature	Date.
Print name	
· · · · · · · · · · · · · · · · · · ·	

17. Adjudication Parties (continued)	
To Broke	
Bar 2 Livestack LLC Iroy Brooks Signature	Date: 4-14-14
Signature	
Troy Brooks	
Print name	
	Date:
Signature	
Print name	
	Deli
Signature	Date:
Print name	
	Date:
Signature	
Print name	
	Data
Signature	Date:
Print name	

17. Adjudication Parties (continued)	
Signature	Date: 4-15-14
Print name	
aure Bignature	Date: 4-15-19
Janice V. Briggs Print name	
Signature	Date: 4-16-14
TOHM BRICES Print name	
Signature	Date:
Print name	
	Date:
Signature	

Print name

Signature aldweek	Date: 4/16/14
HENRY J. CALSUEL JR. Print name	
	Date:
Signature	
Print name	
Signature	Date:
Print name	
	Date:
Signature	
Print name	
Signature	Date:
Print name	

AUSAYMAS CATRE Co. Print name	Date: 4/17/14
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

17. ADJUDICATION PARTIES

Signature Tack Martin	Date: 4/19/14
Tack Martin Print name	
Ady Martin Signature) (Martin	Date: 4-19-14
Print name Signature	Date:
Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

17. ADJUDICATION PARTIES (continued) KIUEZSPRINGS RANCH Date: 4-30-14 Signature MAR SARWGI RANCH Date: 4-30-14 Signature Print name RUERSPRUGI PANCH Date: 4-30-14 Date: Signature Print name

Signature

Print name

Date:_

17. ADJUDICATION PARTIES

Signature Sohn E. Von Schlege// Print name	Date: 5-8-2019
r .	Date:
Signature	
D.i.,	
Print name	-
	Date:
Signature	
Print name	
	Date:
Signature	
Dulin king and the	
Print name	•
<u> </u>	Date:
Signature	
Print name	

() () () () () () () () () ()	
Signature M Col	Date: 4/20/14
Signature	Date: 1/00/)
James M. Root	
James M. Kort	M
Print name	un internationale .
	Date:
Signature	
and Marketing and American State of the Community of the	
Print name	
	Towns in the same of the same of the
en e	
	Date:
Signature	
Print name	
rimenanc	
	Date
Signature	
<u>ran jällistillinilli luun saassa saassa</u>	
Print name	
	Date:
Bignature	
<u> 18 Heritaan (hilita) (halita de la colonia de la colonia</u>	
rint name	

17. ADJUDICATION PARTIES

Jameshi Lost Presidais KBR	Date:
Signature	•
James M Root	
Print name	
	Date:
Signature	
Print name	
C:	Date:
Signature	
Print name	
	,
•	
Signature	Date:
oignature	
	·
Print name	
Signature	Date:
Digitations	
Print name	

17 ADJUDICATION PARTIES

Manage //	Date: 5/10/14
Signature homas Family Ltd. Pa Print name	artnership by Kurt Thomas
	Date:
Signature	
Print name	<u>-</u>
Signature	Date:
Print name	-
Signature	Date:
Print name	_
Signature	_ Date:
Print name	

CONFIDENTIAL

PROPOSED UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT

March 3, 2014

17. Adjudication Parties

Styling Hess Signature	Date: 3-28-14
Steshen Hess Print name	
Signature Physics	Date: 3-28-14
Print name Helling Helling	Date: 3/28/14/
Signature MELISSA HESS Print name	
Signature	Date:
Print name	
Signature	Date:
Print name	

CONFIDENTIAL

PROPOSED UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT

March 3, 2014

Sharen allen Signature SHARON M. AUEN	Date: 3 25 14
Print name	•
Bum & Allen Signature	Date: 3-25-2014
BRIAN E. Allew	
Print name	
Cal ac Aulas	
Signature / Walter Kathleen Gr. Walter	Date: 3-25-2014
Signature /	
Print name	· :
Gen B Lille	Date: 4/16/14
Signature	' (
Cheri & Little	,
Time name	
	•
· · · · · · · · · · · · · · · · · · ·	Date:
Signature	
Print name	

CONFIDENTIAL

PROPOSED UPPER KLAMATH BASIN COMPREHENSIVE AGREEMENT

March 3, 2014

Shaen allen	Date: 3 25 14
Signature SHARON M. AUEN	
Print name	
Buin & allen	Date: 3-25-2014
Signature Brian E. Allew	
Print name	
	Date: 3/28/14
Signature Popson	Date:
Print name	
Ben Hatfild-tyle Signature	Date: 3/28//Y
Becky Hatfield Hyde Print name	
Signature	Date: 3/28/14
Print name	

r" /	
Stany STORICE	Date: 3-10-14
Signature	Date.
·	
Print name	•
·	
Sylvice Richardson	2 12 2016
OGICA NOORO	Date: 3-/2-2017
Signature	
Print name	
Signature Dun	Date: 3-12-2014
Signature	Date.
D.K. Patton DUM	
Print name	

	•
	Date: 3-12-2014
Signature	Date:
Signature Dave Thompson	
FAMOSO CATTLE CO.	
Print name	
Print name	•
Joan Comercal Sea	
LOUNCE COS	Date: 3/12/20/4
Signature	
JOAN AMARAL SERS	
Print name	